Golden Gate University School of Law **GGU Law Digital Commons**

California Agencies

California Documents

4-1-1981

Master Plan for Regulation Review

Office of Administrative Law

Follow this and additional works at: http://digitalcommons.law.ggu.edu/caldocs_agencies



Part of the <u>Administrative Law Commons</u>

Recommended Citation

Office of Administrative Law, "Master Plan for Regulation Review" (1981). California Agencies. Paper 366. http://digitalcommons.law.ggu.edu/caldocs_agencies/366

This Cal State Document is brought to you for free and open access by the California Documents at GGU Law Digital Commons. It has been accepted for inclusion in California Agencies by an authorized administrator of GGU Law Digital Commons. For more information, please contact jfischer@ggu.edu.

California Administrative Register 81, No. 13-Z

SPECIAL EDITION

NOTICE SUPPLEMENT

Master Plan for Regulation Review

APRIL 1, 1981



Published by
OFFICE OF ADMINISTRATIVE LAW
GENE LIVINGSTON, Director

The California Administrative Code Notice Supplement (Z-Register) is a weekly publication comprised of Notices of Proposed Action of State Agencies and a summary of regulations filed with the Secretary of State during a Register week. In addition, any of the following may be included in the publication:

(1) Notices of public meetings of state agencies, including schedules of public meetings or dates by which public comments will be received by agencies during the review of regulations against AB 1111 standards of necessity, authority, clarity, consistency, and

reference.

(2) Orders to repeal any regulations which OAL has determined do not meet the AB 1111 standards cited above.

(3) Executive orders.

(4) Summaries of Attorney General's opinions.

(5) A registry of appointments in state agencies, including vacancies therein.

(6) Miscellaneous notices of general interest to the citizens of California. Subscriptions to the Notice Supplement are available from Publications/Documents

Subscriptions to the Notice Supplement are available from Publications/Documents Section, P.O. Box 1015, North Highlands, CA 95660; telephone: (916) 445-1020. The current subscription rate is \$50.00 per year.



E O LI CO

California launched the Nation's most unique and promising regulatory reform effort when the Legislature passed and the Governor approved AB 1111 (McCarthy). That legislation created the Office of Administrative Law and charged it with the responsibility of conducting a thorough review of all state regulations—new as well as existing ones.

During the first nine months of the Office's existence, the number of new regulations amending or adding to the Administrative Code has been cut by 50 percent. That same critical evaluation is now to be applied to the 30,000 pages of regulations adopted by 123 state agencies during the past four decades.

Regulations affecting all aspects of life—from air pollution to worker health and safety, including coastal protection, energy standards, licensing, and permit requirements—will be judged against the statutory standards of necessity, authority, consistency, reference, and clarity. Any regulation failing to satisfy these standards will be changed or eliminated.

These regulations will first be examined by the adopting agencies in accordance with review plans submitted to the Office of Administrative Law. We have summarized the plans from the 123 agencies into the attached Master Plan as called for by Government Code Section 11349.7. This Master Plan summarizes when and how each state regulatory agency will conduct the review of its existing regulations and describes the mechanism each agency will use to ensure public participation in the review process.

Certainly, we want to encourage active involvement by individuals, groups and businesses as we seek to reduce governmental interference in the lives and enterprises of California residents. Full participation by the public will enhance our efforts to end unwarranted governmental control while maintaining appropriate protection for the economy, the environment, and the health and safety of workers and the public. The purpose of this Master Plan is to aid the public in monitoring and taking part in the comprehensive review of all existing regulations.

GENE LIVINGSTON Director



"Every agency subject to the provisions of this chapter shall undertake a review of all regulations administered by it. . . . The Office of Administrative Law shall, on or before April 1, 1981, file with the Governor, the Senate Committee on Rules and the Speaker of the Assembly a MASTER PLAN FOR REGULATION REVIEW (emphasis added). The Master Plan shall establish specific dates by which each agency shall have complied with this section, including interim review goals to be met

Government Code Section 11349.7

INTRODUCTION

This Master Plan for Regulation Review is submitted pursuant to Government Code Section 11349.7 (d). The purpose of the Master Plan is to provide a schedule of specific dates by which each regulatory agency shall have completed its review of regulations in effect prior to July 1 1980, with the ultimate objective of repealing or amending those which do not meet the statutory standards of necessity, authority, clarity, consistency, or reference.

The Master Plan is also intended to facilitate the understanding of and participation in the

regulation review process by the Legislature and the general public.

Section I provides a general overview of the regulation review requirements of the Adminis trative Procedure Act (AB 1111, McCarthy), the regulatory reform measure signed into law by Governor Brown in 1979.

Section II describes the required contents of each agency plan, the progress agencies have made in developing their plans and the process used by the Office of Administrative Lav (OAL) for approving plans.

Section III contains an outline of the regulation review process, describing required review

activities and OAL's role as monitor and independent reviewer.

Section IV contains a summary of each agency plan, with completion dates and a brie description of the agency's review process, including coordination and public participation mechanisms.

SECTION I: BACKGROUND

The Administrative Procedure Act (APA), Chapter 567 of the Statutes of 1979 (AB 1111 McCarthy), charged all regulatory agencies and the Office of Administrative Law with the orderly review of the almost 30,000 pages of existing government regulations against statutory standards, with the goal of reducing the number and improving the quality of regulations.

The APA requires agencies to develop and submit review plans to OAL, describing how eacl agency will evaluate regulations in effect prior to July 1, 1980 to determine whether such regulations are: (1) necessary to implement, interpret or make specific a law or court decision (2) promulgated by agencies authorized to do so; (3) clearly written so that persons affected by them can easily understand them; (4) consistent with existing laws; and (5) referenced to

a specific statute or court decision.

The original statutory deadline for submitting plans to OAL was July 1, 1980, the date OAI opened its doors. We found at that time that about half of the agencies had prepared review plans. Those that had developed plans had done so without the benefit of guidelines or instruc tions. Consequently, most plans were wholly inadequate. Accordingly, OAL and the Adminis tration worked with the Legislature in amending Government Code Section 11349.7 to extend the deadline for submission of plans to December 31, 1980 (SB 1754, Holmdahl).

In October, the Governor issued Executive Order No. B72-80 instructing all agencies to accelerate the completion of the regulation review process and to ensure that the reviews be conducted with full opportunity for public participation. The Governor's Executive Order required that the review of all existing regulations be completed by December 31, 1982. (A cop) of the Executive Order is attached to this report.) The review is to be completed in accordance

with the following schedule:

June 30, 1981 January 31, 1982 Titles 1, 7, 9, 11, 12, 15, 20, 21, 23 Titles 2, 3, 4, 5, 13, 14, 19 Titles 8, 10, 16, 17, 18, 22, 25 July 31, 1982

Where agencies documented a need for additional time to conduct a thorough review, OAL ha extended the above deadlines on an exception basis.1

Also in October, OAL drafted agency instructions for compiling the review plan. Meetings were held with a number of agencies, legislative staff and several private sector interest groups to obtain their assessment of the draft instructions. Incorporating their suggested changes, the OAL instructions were finalized and sent to agencies on October 24, providing over two months for plan completion.

¹ The review of regulations contained in Title 24 relating to building standards is the responsibility of the Building Standards Commission and thus not include in this schedule.

We project that agencies will complete the review of approximately 10 percent of the regulations by June 30, 1981, 35 percent by January 31, 1982, and 95 percent by July 31, 1982. OAL has approved extension requests beyond July 1982 for the remaining 5 percent of the Administrative Code.

All regulations will have been reviewed well within the original statutory deadline of June 30, 1986. The Office of Administrative Law will conduct its own independent review of those regulations amended or

left unchanged within six months after each agency has completed its review.

Agencies have been encouraged to rely on existing resources to accomplish the review. The Legislature appropriated a total of \$3.5 million annually to be used by agencies and OAL for additional costs associated with review activities. The Governor has ordered that no new funds shall be added to this annua appropriation, directing agencies to meet any additional costs by diverting resources from activities having less priority. Agencies may apply for a portion of the \$3.5 million by submitting a budget change proposal to the Department of Finance, which has the responsibility for allocating the available funds We have been informed that fewer than 25 out of the 123 agencies have requested these funds.

SECTION II: PLANNING FOR REGULATION REVIEW

The OAL instructions for regulation review provided specific guidelines for the developmen of review plans and required that agencies describe how this review process:

(1) would ensure adequate public participation;

(2) would be coordinated with other agencies sharing regulatory responsibilities for program areas which overlap or are closely related;

(3) would be scheduled and staffed to ensure an even flow of activities and the timely

completion of the review.

The statutory deadline for plan submission to OAL was December 31, 1980. Only 65 of the 123 agencies met the deadline, with an additional 41 agencies submitting plans during January

Some plans were received as late as March 24, 1981.

To date, only one agency, the Department of Education, has failed to submit a plan. Despit numerous attempts by OAL to effect compliance, this agency still refuses to follow the mandat of AB 1111. As far as we know, the Department has no intention of submitting a review pla or conducting a review of its regulations. The remedy provided by the Legislature to ensur compliance in this situation is an order to show cause why regulations ought not to be repealed if subsequent to that order, OAL does not receive information supporting the necessity, authority, clarity, and consistency of the regulations, OAL may order the repeal of those regulation which fail to meet one or more of the standards.

OAL staff has closely examined each agency plan and approved those that conformed t statutory mandates and to the Governor's Executive Order. Many plans were not acceptable and required extensive negotiations with regulatory agencies. The majority of agencies, however, the majority of agencies are conformed to the conformed to

er, proved to be cooperative, even though they were late in meeting deadlines.

Issues requiring negotiation related primarily to unacceptable delays in scheduling, inadquate public participation and, in a few instances, to plan provisions that resulted in excessivand unnecessary costs. The great majority of these issues were resolved with agency represent tives. In a number of instances, OAL exercised its statutory authority to set agency schedule and ensure appropriate public participation.

All plans, with whatever amendments were necessary, have now been approved by OAl Many agencies have already begun implementing their plans and a few have even complete

their review of a portion of their regulations.

SECTION III: REGULATION REVIEW PROCESS

The regulatory review process includes the following steps:

1. Agencies develop and OAL approves agency plans for review of all regulations, specifyir sections to be reviewed, dates for public comment and completion of review and the proceused for public participation and coordination.

2. OAL periodically publishes a schedule of reviews based on agency work plans in tl California Administrative Register Notice Supplement (Notice Register). OAL will monit

agency progress and report periodically to the Governor and the Legislature.

3. Agencies conduct reviews, ensuring adequate public involvement. At completion of 1 view, agency submits Statement of Review Completion to OAL. The Statement will be pulished in the Notice Register. For any proposed adoptions, amendments or repeals regulations resulting from the review, agencies must comply with all provisions of Governme Code Section 11346 establishing procedural requirements for regulatory action.

4. For all regulations retained, an agency shall provide to OAL a Statement of Findin summarizing public comment, studies and other reasons or information relied on for its de sion. The Statement of Findings may relate to a cluster or group of regulatory sections, appropriate. In addition to the summary of information, the Statement of Findings shall conta at least the following information: (a) where public comment has been made recommendi the repeal of specific regulations, the agency shall include its reasons for rejecting the pub comment; (b) where regulations that are being retained do not have a specific reference statutory authority, that reference must be cited.

OAT was its own motion or upon petition from a

The Office may request the agency which chose not to repeal a regulation to forward any information on which it relied in reaching its decision. Agencies must respond within 14 calendar days.

- 6. In the event OAL determines through its review that an existing regulation does not meet the standards set forth in Section 11349.1, the Office shall order the adopting agency to show cause why the regulation should not be repealed. Agencies shall respond to OAL no later than 60 days following receipt of the order.
- 7. OAL shall review all information available to it, including the agency's response to the order to show cause and determine whether the regulation meets the statutory standards. If OAL determines that a regulation fails to meet the standards, it shall deliver to the agency, Legislature and Governor a public statement specifying the reasons for its determination.
- 8. Thirty (30) days after delivery of the above statement, OAL shall prepare and transmit an order of repeal of the regulation to the Secretary of State.

Throughout the review process OAL will monitor the activities of the agencies and will publish periodic reports to the Legislature on the progress of each agency. The plan submitted by each agency will serve as the basis for determining progress and adherence to the review mandate. OAL will work closely with each agency to ensure that the plan is carried out in a timely and effective manner.

All those who are interested in the review of regulations are encouraged to participate. To become involved in the review, contact the agency responsible for the regulations concerned.

By referring to the following section which summarizes the plans for each agency, it can be determined who to contact for further information, and the schedule for the review of specific sections of the Administrative Code.

SECTION IV: AGENCY PLANS

On the pages that follow, a summary of each agency's Regulation Review Plan is provided to enable interested parties to become involved in the review process and to determine certain basic information on the conduct of the review.

The following explains how to use the plan summaries.

1. Agencies

The plans are organized by Cabinet level agency, with each department appearing in alphabetical order.

2. Agency Coordinator

The Agency Coordinator is the contact person for the review. Their names, addresses and telephone numbers are provided to enable interested persons to contact agencies for further information and to arrange for participation in the review process.

3. Column Headings

A. Administrative Code Under Review

Under this heading appear the sections or block of sections being considered within a given period of time. Contact the Agency Coordinator to confirm this schedule.

B. Public Comment Ending Date

The date that appears in this column is the final date for receiving public comment by the agency as indicated in the agency plan. Please confirm this with the Agency Coordinator.

C. Statement of Review Completion

This is the date by which the agency will submit its Statement of Review Completion to OAL. The Statement of Review Completion is the document submitted to OAL following an agency's review of all or a group of regulations. The Statement identifies those regulations that the agency had determined do not satisfy the statutory standards and thus will be repealed or amended, as well as those that are found to meet all the standards.

D. Methods of Public Involvement

This column provides a summary of how the agency intends to notify the public of the review process. It is recommended that, if the agency intends to use mailing lists, interested persons ask to be placed on the mailing lists. The public can also request to become involved in advisory groups. Interested parties may wish to subscribe to the Notice Register, which can be accomplished by contacting OAL.

E. Review Process

This summarizes key aspects of the agency's review process. For more information, contact the Agency Coordinator.

F. Coordination

This column lists entities or aspects of coordination to ensure that efforts are made to reduce regulatory inconsistency and duplication. For more information on how agencies intend to carry this out, please contact the Agency Coordinator.

With the publishing of the Master Plan, it is hoped that participation in the review process will be encouraged and facilitated. OAL is convinced that the success of the regulation review effort will be greatly dependent on the extent interested parties and the general public participate. We strongly urge all interested persons and organizations to become involved in this effort.

EXECUTIVE ORDER NO. B72-80

WHEREAS, burdensome, unclear and unnecessary state government regulations increase the business and consumer costs of services and commodities and stifle private initiative and individual responsibility; and

WHEREAS, this Administration is alarmed about the effects of such regulations and is committed to reducing and minimizing the intrusion of government into the affairs of California

businesses and residents; and

WHEREAS, it is the constant obligation of government to reassess its proper role and to seek ways to be more effective and efficient, and particularly so during times of diminishing re-

sources: and

WHEREAS, my signing of AB 1111 makes California the first state to require every state agency to review all regulations with the purpose of reducing their number and improving their quality, and establishes the Office of Administrative Law to direct the completion of the

regulation review by 1986. NOW, THEREFORE, I, Edmund G. Brown Jr., Governor of the State of California, by virtue of the power and authority vested in me by the Constitution and statutes of the State of California, do hereby issue this order to become effective immediately:

1. Agencies reviewing proposed and existing regulations shall eliminate all regulations that do not serve an essential governmental purpose.

2. The review of all existing regulations shall be accelerated and completed by December 31, 1982 provided, however, that an agency may seek from the Office of Administrative Law a later completion

date if extraordinary circumstances would render completion by December 31, 1982 impossible.

3. State agencies shall insure that representatives of persons who are affected by their regulations are effectively involved in the review of all existing regulations. The Office of Administrative Law shall monitor agencies to ensure public participation in the review of existing regulations.

4. No new funds shall be added to the state budget for the accelerated review of regulations. Each year costs for conducting the review of existing regulations shall be allocated from an annual appropriation amount not to exceed the amount of the 1980–81 Budget Act for regulation review. Additional staffing

needs shall be met by diverting existing resources.

5. The Office of Administrative Law shall deliver periodic reports to the Governor and the Legislature summarizing each agency's performance in meeting the goal of reducing the number of regulations and

improving their quality.



EDMUND G. BROWN IR. Governor of California ATTEST: MARCH FONG EU Secretary of State MICHAEL S. GAGAN Deputy Secretary of State IN WITNESS WHEREOF I have hereunto set my hand and caused the Great Seal of the State of California to be affixed this 9th day of October

MASTER PLAN FOR REGULATION REVIEW BY AGENCY

April 1, 1981

•

Administrative Code Title	Regulatory Agency	Page
	BUSINESS, TRANSPORTATION AND HOUSING AGENCY	
2	Administer of Local Agency Security	1
4	Alcoholic Beverage Control Appeals Board	2
$\bar{4}$	Alcoholic Beverage Control, Department of	3
10	Banking, State Department of	4
$\overline{21}$	Business, Transportation and Housing Agency	5
13	California Highway Patrol	6
10	Corporations, Department of	8
25	Housing and Community Development, Department of	9
25	Housing Finance Agency, California	11
10	Insurance, Department of	12
13	Motor Vehicles, Department of	13
13	New Motor Vehicle Board	14
10	Real Estate, Department of	15
10	Real Estate, Department of	16
10	Small Business Development, Office of	17
4/21	Transportation, Department of	18
	HEALTH AND WELFARE AGENCY	
9	Alcohol and Drug Programs, Department of	19
9/17	Developmental Services, Department of	21
22	Employment Development Department	23
22	Health and Welfare Agency	24
17/22	Health and Welfare Agency	26
9	Mental Health, Department of	30
9	Rehabilitation, Department of	32
22	Social Services, Department of	34
22	Statewide Health Planning and Development, Office of	42
22	Unemployment Insurance Appeals Board, California	44
	INDUSTRIAL RELATIONS	
8	California Occupational Safety and Health Administration	45
18	Industrial Relations, Department of	47
8	Industrial Welfare Commission	49
	RESOURCES AGENCY	
14	Boating and Waterways, Department of	50
14	California Coastal Commission	51
14	Coastal Conservancy, State	
14	Colorado River Board	53
14	Conservation, Department of	54
14	Fish and Game, Department of	
14	Forestry, Department of State Board of Forestry	59
14	Parks and Recreation, Department of	63
23	Reclamation Board	
14	Resources Agency	
14	San Francisco Bay Conservation and Development Commission	67
14	Solid Waste Management Board	
23	Water Resources Control Board	71
1/23	Water Resources, Department of	75
	STATE AND CONSUMER SERVICES AGENCY	
16		77
10	Accountancy, Board of	78
16	Architectural Examiners, Board of	79
4	Athletic Commission, State	
16	Automotive Repair, Bureau of	81
16	Barber Examiners, Board of	82
16	Behavioral Science Examiners, Board of	83
16	Cemetery Board	
16	Certified Shorthand Reporters, Board of	
16	Collection and Investigative Services, Bureau of	
16		
16	Construction Inspectors, Board of Registered	00
16	Cosmetology, Board of	
16	Dental Examiners Roard of	Of

Administrative Code Title	Damilaham, Amanan	Page
		_
16 16	Electronic and Appliance Repair, Bureau of Employment Agencies, Bureau of	91 92
16	Fabric Care, Board of	
2	Fair Employment and Housing Commission	94
19	Fire Marshal, State	95
18	Franchise Tax Board	97
16	Funeral Directors and Embalmers, Board of	102
2/21	General Services, Department of	103
16	Geologists and Geophysicists, Board of Registration for	106
16	Guide Dogs for the Blind, Board of	107
4	Home Furnishings, Bureau of	108
4	Horse Racing Board, California	109
16	Landscape Architects, Board of	111
16	Medical Quality Assurance, Board of	112
2	Museum of Science and Industry	114
16 16	Nursing, Board of Registered	110
16	Nursing Home Administrators, Board of	117
2	Optometry, Board of	110
15	Pharmacy, Board of	
16	Professional Engineers, Board of Registration for	120
2	Public Employees' Retirement System	121
$1\overline{6}$	Structural Pest Control Board	122
16	Tax Preparer Program	123
5	Teachers' Retirement System, State	124
12	Veterans Affairs, Department of	125
16	Veterinary Medicine, Board of Examiners in	126
16	Vocational Nurse and Psychiatric Technician Examiners, Board of	127
	YOUTH AND ADULT CORRECTIONAL AGENCY	
15	Corrections, Board of	128
15	Corrections, Department of	
15	Narcotic Addict Evaluation Authority	130
15	Prison Terms, Board of	
15	Youth Authority, Department of the	132
15	Youthful Offender Parole Board	133
	OTHER AGENCIES	
8	Agricultural Labor Relations Board	134
13/17	Air Resources Board	
2	Arts Council, California	
16	Chiropractic Examiners, Board of	140
5	Community Colleges, California	141
2	Control, Board of	. 143
4	Educational Facilities Authority, California	147
20	Energy Commission, California	. 148
18	Equalization, Board of	149
2	Fair Political Practices Commission	
3/4	Food and Agriculture, Department of	. 152
4	Health Facilities Commission, California	. 163
25	Housing Bond Committee	
2	Lands Commission, State	
5 5	Library, California State	
16	Maritime Academy, California	168
11	Peace Officer Standards and Training, Commission on	160
7	Pilot Commissioners for the Bays of San Francisco, San Pablo and	. 103
•	Suisun, Board of	. 170
8	Public Employment Relations Board	
19	Seismic Safety Commission	. 172
5	Student Aid Commission	. 173
5	Teacher Preparation and Licensing, Commission for	. 175
21	Transportation Commission	. 176
5	University and Colleges California State	177

(Register 81, No. 13-Z-4-1-81)

Administrative Code Title	Regulatory Agency	Page
	CONSTITUTIONAL OFFICERS	
5	Education, Department of	179
2/18	Controller's Office, State	182
11	Justice, Department of	184
2	Secretary of State	186
2	Treasurer, State	187
10	Districts Securities Division	188
4	California Pollution Control Financing Authority	190
	GOVERNOR'S OFFICE	
19	Emergency Services, Office of	191
14	Planning and Research, Office of	192

AGENCY: Administrator of Local Agency Security

AGENCY COORDINATOR: James F. Carrig

ADDRESS: 235 Montgomery St., Suite 750, San Francisco, CA 94104

TELEPHONE NUMBER: (415) 557-0183

OAL Master Plan For Review of Existing Regulations

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2					
Division 4.5 Local Agency Deposit Security	June 81		The Weekly Bulletin of the State Banking Department will be utilized to publicize the review process. It will be mailed out to all persons on the Department's mailing list. In addition, mailings will go to all state-chartered and federal savings and loan associations in California.	The regulations of the Administrator of Local Agency Security will be reviewed in much the same manner as the Superintendent of Banks. Each unit of review will be specified in the Weekly Bulletin. A 60-day comment period will be provided for each unit. Internal Task Forces will be established for the review.	As required.

AGENCY: Alcoholic Beverage Control Appeals Board

AGENCY COORDINATOR: Floyd Mitzner

ADDRESS: 1215 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4005

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 1.1 Alcoholic Beverage Control Appeals Board	Apr. 81	May 81	Trade journals	The Alcoholic Beverage Control Appeals Board regulations govern appeals procedure.	Department of Alcoholic Beverage Control
				The review will be carried out by staff.	
·				Comments from the public are welcome.	

AGENCY: Department of Alcoholic Beverage Control

AGENCY COGEDINATOR: Jerry Whitfield

ADDRESS: 1215 0 Street, Sacramento, CA 95814

TELEPHON® NUMBER: (415) 881-3951

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF ; PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 1 Department of Alcoholic Beverage Control	Sept. 81	Dec. 81	Mailing List Newspapers Trade journals Public service announce- ments on radio and TV	Preliminary internal and final reviews will be made by a select 10-member project committee. Issue papers will be developed which will include committee recommendations.	Board of Equalization Collection of Excise Taxes for Alcoholic Beverages; §§ 7,8,54, 68.2,71,79 Department of Consumer Affairs and Department of Health Services Labeling and Advertising; §§ 80,103, 104,130 Department of Finance Population Estimates § 61.3 Alcoholic Beverage Control Appeals Board
	·				
		·			

AGENCY: State Banking Department

AGENCY COORDINATOR: James F. Carrig

ADDRESS: 235 Montgomery St., Suite 750, San Francisco, CA 94104

TELEPHONE NUMBER: (415) 557-0183

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
UNDER REVIEW Title 10 Subchapter 10 Banks Subchapter 50 Business and Industrial Development Corporations Subchapter 1 General Provisions Subchapter 5 Administration	May 81 Dec. 81 Jan. 82 Jan. 82	March 82 May 82 July 82 July 82	The Departmental Weekly Bulletin will be used to publicize the review pro- cess, and will be sent out to all persons on the Departmental mailing list. A Task Force composed of industry representatives, independent experts and Department representatives will be established for each unit of review.	A 60-day comment period will be provided for units of regulations under review. The Task Forces established will recommend changes and review drafts of changes. Task Forces will meet monthly at the outset.	As required.

AGENCY: Business, Transportation and Housing Agency

AGENCY COORDINATOR: Judith Nelson

ADDRESS: 1120 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-1321

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 21					
Chapter 3 Subchapter 1 Business Transportation Demonstration Program	May 81	June 81	Subchapters 1, 5 Newspaper Advertisements Mailing Notices to: Transportation planning agencies and county	The Business, Transporta- tion and Housing Agency is supported by the Depart- ment of Transportation for the review.	California Housing Finance Agency Department of Veterans Affairs Department of Savings
Subchapter 5 Allocation of Funds for Special Public Transportation Needs	May 81	June 81	transportation commis- sions, county auditors, county administrative officers, mayors, trans- it claimants of Transpor- tation Development Act		and Loan State Banking Department Department of Corporations
Subchapter 2 Transportation Development	Apr. 81	June 81	funds and other interest- ed parties. Subchapters 2, 2.5		Department of Insuranc Department of Real Estate
Subchapter 2.5 State Transit Assistance Program	Apr. 81	June 81	Public hearings on these regulations were sched- uled for January 29,		
Subchapter 4 Regulations Pursuant to the Housing Financial Discrimination Act of 1977	Apr. 81	June 81	1981 and February 25,26, 27, 1981. Subchapter 4 Newspapers Mailings to Financial Institutions		
Articles 1-6					

Page 6

AGENCY: California Highway Patrol

AGENCY COORDINATOR: Herb Yarbrough

ADDRESS: P.O. Box 898, Sacramento, CA 95804

TELEPHONE NUMBER: (916) 322-7157

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 13 Chapter 2			Newspapers Mailing Lists	Review of existing regula- tions is being accomplished	Department of Motor Vehicles
Subchapter 1 Licensed Stations and Muffler Instal-			Notice Register Advisory Committees	in conjunction with ongoing regulatory process.	Department of Health Services
lers Articles 1, 2	Apr. 81	July 81		Notices are published regu- larly in newspapers and Notice Register	Department of Education Office of Traffic Safet Department of Transpor-
Article 3	Sept. 81	Nov. 81		Notice Register	tation Office of the State
Subchapter 2 Lighting Equipment	Apr. 81	Sept. 81	4		Fire Marshal
Subchapter 4 Special Equipment					
Articles 1,3,4,8, 13	Apr. 81	Sept. 81			
Article 2	May 81	July 81			
Articles 9,10,11	Apr. 81	July 81			
Article 12	May 81	Aug. 81			
Article 14	July 81	Oct. 81			
Article 15	Aug. 81	Nov. 81			
Subchapter 5 Special Vehicles					
Articles 2,5	May 81	Aug. 81			
Article 3	Apr. 81	Sept. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 6 Hazardous Material	Sept. 81	Nov. 81			
Subchapter 6.5 Motor Carrier Safety	Sept. 81	Nov. 81			
Subchapter 7 Loading Regulations		Marine and Property and Propert			
Article l	Oct. 81	Dec. 81			
Article 2	Apr. 81	Aug. 81			
Articles 3,4,5,6	May 81	Sept. 81			
Article 7	May 81	Oct. 81			
Articles 8,9	June 81	Oct. 81			
Articles 10, 11	June 81	Nov. 81			
Subchapter 9 Abandoned Vehicles	Sept. 81	Nov. 81			
Subchapter 10 Internal Affairs	Mar. 81	May 81			
	ng pagaga di di				
		u de la companya de			

AGENCY: Department of Corporations

AGENCY COORDINATOR: R. E. La Noue,

ADDRESS: 1025 P Street, Rm. 205, Sacto, CA 95814

TELEPHONE NUMBER: (916) 322-3633

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 10					
Subchapter l General Provisions	Dec. 81	Feb. 82	Newspapers Mailing Lists	At the outset of each phase of the review,	Department of Health Services
Subchapter 2 Corporate Securities	Dec. 81	Feb. 82	Departmental Licensees Corporate Securities Law	written public comments will be solicited. Office of Policy Staff Attorneys	
Subchapter 2.6 Franchises	Dec. 81	Feb. 82	Advisory Committee, Knox-Keene Health Care Service Plan Act Advisory	will evaluate public comments/suggestions and	
Subchapter 3 Credit Unions	May 82	June 82	Committee, Credit Union Law Advisory Committee,	staff reviews and will propose changes. Depart-mental Executive Staff will review proposed changes.	
Subchapter 4 Industrial Loan Law	May 82	June 82	Industrial Loan Law Advisory Committee, Personal Property Broker Law Advisory Committee		
Subchapter 5.5 Health Care Service Plans	May 82	June 82			
Subchapter 6 Personal Property Brokers	May 82	June 82			
Subchapter 7 Security Owners Protection	May 82	June 82			
Subchapter 10 Check Sellers and Cashers	May 82	June 82	,		,
Subchapter 11 Security Deposi- tories	May 82	June 82			
			į.		

AGENCY: Department of Housing and Community Development

AGENCY COORDINATOR: Jean Arimoto

ADDRESS: 921 Tenth Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4775

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
UNDER REVIEW Fitle 25 Chapter 7 Department of Housing and Community Development Programs Subchapters 1, 1.5, 2,3,4,5,6,7,8,10, 11 Chapter 6.5 Program Operations Subchapter 1 Subchapter 2 Subchapter 3 Chapter 2		Mar. 81 June 81 July 81	Newsletter Mailing List composed of city managers, county administrators, Legislators, redevelop- ment agencies, community groups, and interested individuals Newspapers Notice Register Housing and Community Development Commission Meetings Public Meetings	Public hearings will be scheduled; for further information contact Coordinator.	Building Standards Commission California Housing Finance Agency
Mobilehome Parks Act					
Subchapter l	Sept. 81	Nov. 81			
Subchapter 2	July 81	Aug. 81			The state of the s
Chapter 1 State Housing Law Regulations and Earthquake Protection Law Regulations			,		,
Subchapter 3	July 81	Sept. 81			
Subchapter 1	Sept, 81	Nov. 81			

Page <u>10</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 3 Factory-Built Hous- ing and Mobilehomes					
Subchapter 1	May 81	June 81			
Subchapter 2	Nov. 81	Jan. 82			
Subchapter 4	Apr. 81	July 82			
Chapter 6 Department of Hous- ing and Community Development Programs					
Subchapter 1	Aug. 81	Sept. 81			·
Subchapter 3	Aug. 81	Sept. 81			
Subchapter 4	Aug. 81	Sept. 81	1		
Subchapter 5	Oct. 81	Nov. 81			
					,

AGENCY: California Housing Finance Agency

AGENCY COORDINATOR: Roger A. Clay

ADDRESS: 301 Capitol Mall, Suite 403, Sacto, CA 95814

TELEPHONE NUMBER: (415) 557-2740

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 25					
Chapter 1 Conflict of Inter- est	May 81	Sept. 81	Mailing List	Comments will be solicited for discussion at meetings of the Board of Directors.	Business, Transportation and Housing Agency Department of Housing
Chapter 2 Rental Housing	June 81	Sept. 81			and Community Develop- ment State Treasurer
Chapter 3 Single Family MPP	July 81	Sept. 81			State Treasurer
Chapter 4 Board Procedures	Aug. 81	Dec. 81			
Chapter 5 Home Ownership	Sept. 81	Dec. 81			
Chapter 6 Insurance	Oct. 81	Dec. 81			

AGENCY: Department of Insurance

AGENCY COORDINATOR: Leo M. Hirsch

ADDRESS: 100 Van Ness Avenue, 17th Floor, San Francisco, CA 94102

ATSS 597-3908

TELEPHONE NUMBER: (415) 557-3908

Page <u>12</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 10 Chapter 5 Subchapter 1 Production of Insurance Subchapter 2 Policy Forms and Other Documents Articles 6.7 and 6.9 Credit Life and Disability Policy Forms and Rates Subchapter 3 Insurers Subchapter 4 Insurer Securities Permits Subchapter 5 Stock Insurers— Corporate Affairs	Jan. 82 Nov. 81 Oct. 81 May 82 Feb. 82 Oct. 81	June 82 July 82 Dec. 81 July 82 Apr. 82 Feb. 82	Subchapter 1: Meetings with Joint Producers' Council composed of: California Association of Life Underwriters Professional Insurance Agents Independent Insurance Agents Association of California Insurance Brokers Society of Southern California Society of Insurance Brokers Western Association of Insurance Brokers Western Association of Insurance Brokers Subchapters 2 and 3: Notices will be sent to persons on the Bulletins and Rulings subscription list Newspapers Press Releases Trade Journals Subchapter 4: Public Meeting	The Department will utilize all input in determining what recommendations should be made with respect to the regulations reviewed. This includes review of written comments, suggestions pursuant to notices for comments, review of transcripts of public meetings and recommendations of staff.	Commissioner

AGENCY: ___

Department of Motor Vehicles

AGENCY COORDINATOR:

Madeline Rule

ADDRESS:

P.O. Box 1828, Sacramento, CA 95814

TELEPHONE NUMBER:

(916) 445-6601

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 13 Chapter 1 Subchapter 1 Article 1 Conflict of Interest Code Article 2 Drivers Licenses Article 3 Registration	Mar. 81 Feb. 81 Dec. 81	Mar. 81 Mar. 81 Dec. 81	Mailing Lists Notice Register Newspapers	Preliminary analysis of existing regulations will be conducted by DMV staff. Issue papers including proposed revisions will be prepared based on this preliminary analysis. These papers will be reviewed by legal staff and other concerned agencies. Public comments will be	Department of Boating and Waterways
Article 3.1 Registration of Vessels Article 3.2 Notice Procedures for Seizure of Vehicles	Dec. 81	Dec. 81		considered and those accepted will be incorporated into the final regulatory changes.	·
Article 3.5 Procedures for Requesting Registration Information Article 4 Occupational Licenses	Dec. 81	Dec. 81	·		

AGENCY: New Motor Vehicle Board

AGENCY COORDINATOR: Sam W. Jennings

ADDRESS: 1401 21st St., Ste. 407, Sacto, CA 95818

TELEPHONE NUMBER: (916) 445-1888

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 13					
Chapter 1			Newspaper	Review process began in	Department of Motor
Subchapter 2			Mailing List Regulation Advisory Commit-	March. Public comment is invited, and a hearing	Vehicles
Article 1 Administration (§8 551.9,551.10)	Mar. 81	Apr. 81	tee consisting of one public member and one auto dealer member of the	will be scheduled in October 1981 to consider	
Article 6 Form (8597.1)	Mar. 81	Apr. 81	Board Public Meetings		
Article l Administration	Oct. 81	Jan. 82			·
Article 2 Filing of Peti- tion	Oct. 81	Jan. 82	·	·	·
Article 3 Appeals From Decisions of the Department	Oct. 81	Jan. 82			
Article 4 Hearings on Peti- tions and Appeals		Jan. 82			
Article 5 Protests	Oct. 81	Jan. 82			
Article 6 Form	Oct. 81	Jan. 82			
Protests Article 6					

AGENCY: Department of Real Estate

AGENCY COORDINATOR: Marty Dingman

ADDRESS: P.O. Box 160009, Sacramento, CA 95816

TELEPHONE NUMBER: (916) 323-2139

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC STATEMENT COMMENT ENDING DATE COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Division 5 Chapter 6 Real Estate Commissioner Articles 1-25 Sections 2700-3013	Jan. 82 Apr. 82	Mailing List Newspaper Notices Advisory Committees: Advisory Commission, Subdivision Advisory Committee, Research and Education Advisory Committee, composed of interested industry groups and public at large.	Regulations will initially be reviewed by staff. Advisory committees will then review and propose additional changes. Public written comment will be solicited from mid-November 1981 to mid-January 1982.	Department of Consumer Affairs Department of Corpor-

AGENCY: Department of Savings and Loan

AGENCY COC RDINATOR: Alec C. Stein

ADDRESS: 350 Sansome Street, San Francisco, CA 94104

TELEPHONE NUMBER: (213) 736-2801, ATSS 677-2801

ADMINISTRATIVE CODE UNDER REVIEW	COMMENT	STATEFENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 10 Chapter 2 Savings and Loan Commissioner Subchapters 1,1.3 1.5,2,3,4,4.5,5,6 Subchapters 7,7.5,7.6,8,9,10,11,12,13,14,15,16,17,18,19,20,21,22,26	Mar. 82	July 82	Mailing list; includes all state licensed savings and loan associations, attorneys, public interest law firms, economists, accountants, consumers, and media representatives.	Written comments on proposed changes, additions, amendments or deletions from existing regulations will be solicited from the mailing list, with the Executive Committee overseeing and evaluating the regulation review.	Business, Transportation and Housing Agency

AGENCY: Office of Small Business Development

AGENCY COORDINATOR: Clarence Williams

ADDRESS: 1120 N Street, Room 2117, Sacto, CA 95814

TELEPHONE NUMBER: (916) 445-6545

OAL Master Plan For Review of Existing Regulations

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 10 Chapter 7 Job Development Corporation Law Executive Board Sections 5000- 5077	July 81	Sept. 81	Newspapers Notice to Urban and Rural Development Corporations	Small Business Development Board of Directors composed of Superintendent of Banks, Director of the Department of Economic and Business Development, Director of the Department of Food and Agriculture and nine members appointed by the Governor will review proposed changes	

AGENCY: Department of Transportation

AGENCY COORDINATOR: Richard East

ADDRESS: 1120 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4860

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 6 Outdoor Advertising			Newspaper Advertisements Mailings to Organizations,	It is anticipated that as the review process iden-	As required.
Subchapters 1,2,3, 4,5	Oct. 81	Dec. 81	Political Subdivisions and Individuals	tifies regulations which require amendment or re- peal, regulatory proceed-	
Title 21				ings will be initiated.	
Chapter 2 Department of Trans- portation				·	
Subchapters 1,2,3, 4,5,6,7,9,10,11, 12,13,14,15,18,19	May 81	June 81			
Chapter 2.5 Division of Aero- nautics					
Subchapters 1,2, 2.1,2.5,4,4.5,6	Apr. 81	June 81			

ATOR: Penny Parker

AGENCY COORDINATOR: Penny Parker

ADDRESS: 111 Capitol Mall, Sacramento, CA 95814

AGENCY: Alcohol and Drug Programs, Department of

TELEPHONE NUMBER: (916) 323-2064

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLITION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 9					
Chapter 1 Department of Mental Health			Mailing List Newspapers Meetings with Advisory	After completion of review, regulatory hearings will be held during August and	Services Department of Mental
Subchapter 3 Community Mental Health Services (§§ 781-788, 790)	May 81	June 81	Groups	September 1981. Time and location have not yet been determined.	Health Department of Social Services Department of Motor Vehicles
Subchapter 4 Community Mental Health Services (\$824Facilities for Inebriates and Alcoholics)	May 81	June 81			
Subchapter 6 Methadone Pro- grams	May 81	June 81			
Subhcapter 7 Methadone Drug Analysis Labor- atories	May 81	June 81			
Subchapter 8 Recovery Houses	May 81	June 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 4 Department of Alcohol and Drug Abuse					
Subchapter 1 Administration	May 81	June 81			
Subchapter 2 Development and Administration of County Alcoholism Programs	May 81	June 81		,	
Subchapter 3 Drinking Driver Program	May 81	June 81			
				·	
			·		

Page 21

AGENCY: Department of Developmental Services

AGENCY COORDINATOR: Joan M. Nelson

ADDRESS: 744 P Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 492-7082

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 9					
Chapter 1			Mailing List	After the review process,	Department of Mental
Subchapter 4 Community Mental Health Services Under the Short- Doyle Act	Apr. 81	May 81	Advisory Groups	regulatory hearings will be held as necessary.	Health Department of Health Services
Subchapter 5 Community Mental Health Services Under the Lanter- man-Petris-Short Act	Apr. 81	May 81			
Title 17					
Part II					
Chapter 1 Mental Retardation Regional Centers	July 81	Jan. 82			
Chapter 2 Parental Reimburse- ment for Services	July 81	Dec. 81			
Chapter 3 Operational Stand- ards for Regional Centers					
Sections 50301- 50327	Aug. 81	Jan. 82			

Page 22

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Sections 54000- 54010	June 81	Dec. 81			
Chapter 4 General Provisions					
Subchapter 5	June 81	Aug. 81			
Subchapter 6 Fiscal Audit Appeals	Nov. 81	May 82			·
*New Subchapter 5 scheduled to be adopted in Aug. 81					
í	.			,	3

AGENCY: Employment Development Department

AGENCY COORDINATOR: Ron Fox

ADDRESS: 800 Capitol Mall, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-0325

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC STATEME COMMENT OF ENDING DATE COMPLET	METHODS OF	REVIEW PROCESS	COORDINATION
Title 22 Division 1 Part 1 Unemployment Compensation Chapters 1 and 2 Sections 125-1 455.5-13 Chapters 3 and 4 Sections 606-1 1181-2 Chapters 5-10 Sections 1251-1- 2113-1 Parts 2-4 Disability Compensation and Extended Unemployment Compensation Division 3 Employment Services Programs Sections 2601-1 - 4900-11	April 81 May 81 July 81 Aug. 83 Oct. 81 Nov. 83 Nov. 81 Jan. 8	Newspaper Mailing Lists used to distribute issue papers Public Hearings California Administrative Register	Public hearings are scheduled in Sacramento on: April 10, 1981 July 10, 1981 October 9, 1981 December 18, 1981	As required.

Page 24

AGENCY: Health and Welfare Agency

AGENCY COURDINATOR: Bill Wilder

ADDRESS: 915 Capitol Mall, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 323-4792

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 22 Division 8 Chapter 1 Purpose, General Definitions and General Provisions	June 81	July 81	Agency Mailing List	The Health and Welfare Agency has developed an explanatory summary which is available upon request.	Fair Employment and Housing Commission Other State Agencies as required
Chapter 2 Discriminatory Practices Relating to All Groups Protected by Article 9.5	June 81	July 81			
Chapter 3 Discriminatory Practices Relating to Specific Groups Protected by Article 9.5		July 81			
Chapter 4 Compliance and Enforcement	June 81	July 81			
Chapter 5 Relationship Be- tween the Secre- tary, Department of Fair Employment and Housing and	June 81	July 81	·		

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Fair Employment and Housing Commission					
Sections 98000- 98413	June 81	July 81			

Page 26

AGENCY: Department of Health Services

AGENCY COORDINATOR: Ron Wetherall

ADDRESS: 714 P Street, Room 1650, Sacto, CA 95814

TELEPHONE NUMBER: (916) 322-4990

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 17 Public Health Chapter 1 Administration Chapter 2 Laboratories Chapter 3 Local Health Service Chapter 3.1 Indian Health Chapter 4 Preventive Medical Service Chapter 5 Sanitation Subchapter 1 Engineering Water Only Subchapter 1.5 California Environmental Quality Act Subchapter 2 Food and Drugs Subchapter 3 Vector Control Subchapter 3.1 Wild Animals	July 82 Mar. 84 Mar. 84 Aug. 81 Mar. 84 Sept. 81 July 81 Mar. 84 Sept. 83 Sept. 83	Oct. 82 June 84 June 84 Dec. 81 June 84 June 84 Jan. 82 Nov. 81 June 84 Dec. 83 Dec. 83	Notices Newspapers Department's Mailing List Supplemental Mailing Lists where needed News Release	The Department will seek the broadest public participation possible through the use of public notices, newspapers, the Department's mailing list, news releases and supplemental mailing lists where needed. Ad hoc committees made up of both the regulators and the regulated will be convened for participation in evaluating proposals when needed.	The Department has identified 24 regulatory agencies with which it proposes to coordinate the AB 1111 review process. Among those agencies so identified are the Departments of Aging, Alcohol and Drug Programs, Developmental Services, Education, Rehabilitation, Social Services, the Employment Department, and the Office of Statewide Health Planning and Development.

AUMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT EMDING DATE	STATEMENT OF COMPLETION	METHODS OF . PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 4 Radiation Subchapter 4.5 Radiologic Tech-	Oct. 81	Feb. 82			
nology Subchapter 5 Air Sanitation Subchapter 6	Sept. 81	Jan. 82 Dec. 83			
Organized Camps Subchapter 7 Wiping Rags	Sept. 83	Dec. 83			
Subchapter 8 Drinking Utensils	Sept. 83	Dec. 63			
Title 22 Division 2 Conflict of Interest Code	July 81	Nov. 81			
All Others	Aug. 81	Dec. 81	,	The state of the s	
Division 3					
Chapter 1 Introduction	July 81	Nov. 81			
Chapter 2 Determination of Medi-Cal Eligi- bility and Share of Cost	July 81	Nov. 81			
Chapter 3 Health Care Services	Oct. 81	Feb. 82			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 3.5 Pilot and Demon- stration Projects	Aug. 81	Dec. 81			
Chapter 4 Pre-paid Health Plans	Aug. 81	Dec. 81			
Chapter 5 Adult Day Health Care	Sept. 81	Jan. 82			
Chapter 11 Drug Formulary and Medical Supplies Listing	Oct. 81	Feb. 82			
Division 4 Environmental Health					
Water	Sept. 81	Jan. 82			
Hazardous Wastes	Oct. 81	Feb. 82			
Division 5					
Chapter 1 General Acute Care Hospitals	May 82	Sept. 82			
Chapter 2 Acute Psychiatric Hospitals	Mar. 84	June 84			
Chapter 3 Skilled Nursing Facilities	May 81	Sept. 81		·	

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 4 Intermediate Care Facilities	Mar. 84	June 84			
Chapter 5 Referral Agencies	Sept. 82	Dec. 82			
Chapter 6 Home Health Agencies	Oct. 81	Feb. 82			
Chapter 8 Intermediate Care Facilities for the Developmentally Disabled	Mar. 84	June 84			
Chapter 10 Adult Day Health Centers	Sept. 81	Jan. 82			

AGENCY: Department of Mental Health

AGENCY COORDINATOR: Ellen Goldman

ADDRESS: 2260 Parktowne Circle, Sacramento, CA 95825

TELEPHONE NUMBER: (916) 920-7164

OMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
	June 81 June 81 June 81			After internal review, public comments will be solicited. Staff will prepare final recommendations which will be reviewed by the departments legal staff.	Department of Health Services Department of Alcohol and Drug Abuse Department of Social Services Department of Developmental Services Department of Corrections California Health Facilities Commission Department of Aging Department of Rehabilitation Office of Statewide Health Planning and Development Department of Consumer Affairs Department of Youth Authority State Public Defenders Office

Page 31

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 9 Audit Appeals of Community Mental Health Services Under the Short-	June 81	Sept. 81			
Doyle Act					
	·				
					·

AGENCY: Department of Rehabilitation

AGENCY COORDINATOR: Joe Calderon

ADDRESS: 830 K Street Mall, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 323-2243

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 9					
Chapter 3 Department of Rehabilitation			Newspapers Mailing Lists Rehabilitation Advisory	The review process will occur in four stages: review by advisory commit-	As required.
Subchapter 1 Authority	Mar. 81	Apr. 81	Committee Advisory Committee for Services for Blind and	tees, internal review by departmental staff, solicitation of public comment	
Subchapter 2 Declaration of Purpose	Mar. 81	Apr. 81	Partially Sighted Counselor Advisory Committee	and public meetings. Public meetings are scheduled in Sacramento on	
Subchapter 3 General Provisions	Mar. 81	Apr. 81	Vocational Rehabilitation Appeals Board Advisory Committee for	March 2 and 3, 1981 and in Los Angeles on March 5 and	
Subchapter 4 Definitions and Terms	Mar. 81	Apr. 81	Services to the Deaf and Hearing Impaired Rehabilitation Facilities	6, 1981.	
Subchapter 5 Scope of Services	Mar. 81	Apr. 81	Advisory Committee		
Subchapter 6 Confidentiality	Mar. 81	Apr. 81			·
Subchapter 8 Authorization of Services	Mar. 81	Apr. 81			
Subchapter 9 Rates of Payment	Mar. 81	Apr. 81	,		

					Page 33
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 10 Rehabilitation Facilities and Community Resource for the Handicap- ped	Mar. 81	Apr. 81			
Subchapter 11 Administrative Review and Fair Hearings Procedure	Mar. 81	Apr. 81			
Subchapter 14 Conflict of Interest Code	Mar. 81	Apr. 81			
				·	
		-			

AGENCY: Department of Social Services

AGENCY COORDINATOR: Robert C. James

ADDRESS: 744 P Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-0384

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 22 Division 2. Adoptions Chapter 1 General Provisions Division Chapter 2 Licensing Agencies Chapter 3 Adoptions & Child Placing Agency Chapter 4 Minimum Standards for Facilities for Children Division 6. Licensing Chapter 1 General Requirements Chapter 2 Residential Facilities Chapter 3 Adult Day Care Facilities Chapter 4 Homefinding and Adoption Agencies	Sept. 81 Jan. 82 Sept. 81 May 81	Oct. 81 Oct. 81 Mar. 82 Oct. 81 July 81 July 81 June 81 June 81	The Department will use a variety of methods to solicit public participation. These include newspapers, mailing lists, all-county letters. In some instances, task forces or advisory groups will be formed. Please contact the agency coordinator to obtain further details regarding public involvement.	The Department will develop issue papers, establish task forces, and hold public hearings as required.	Department of Aging Department of Health Services Department of Developmental Services Department of Rehabilitation State Council on Developmental Services Department of Education Department of Mental Health Office of Statewide Health Planning and Development Employment Development Department California Youth Authority State Fire Marshal Department of Justice Alcohol and Drug Programs County Welfare Departments Building Standards Commission

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 5 Continuing Care	Sept. 81	Oct. 81			
Chapter 6 Foster Family Homes	May 81	July 81		•	
Chapter 7 Standards for Family Day Care Facilities	Apr. 81	July 81			
Manuals of Policies and Procedures					·
Division 02 Organization and Functions of Depart- ment of Social Services	Sept. 81	Oct. 81			
Division 06* Foster Family Home	Sept. 81	Oct. 81			
Division 10 Administrative Standards for Public Social Services	June 81	July 81			
Division 11 Administrative Standards for Eligi- bility and Assistance Programs	Sept. 81	Oct. 81			
Division 14 Staff Development and Training	Apr. 82	July 82			
*New division to be					

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Division 15					
Chapter 000 Eligibility Control	Sept. 81	Oct. 81			
Chapter 100 General	Apr. 81	Apr. 82			
Chapter 200 Organization for Quality Control/ Corrective Action	Sept. 81	Oct. 81			
Chapter 300 The Quality Control/ Corrective Action Process	Apr. 81	Apr. 82			
Chapter 400 Quality Control/ Corrective Action Reporting	Sept. 81	Oct. 81			
Division 17 Department Communi- cations	Sept. 81	Oct. 81	·		
Division 19 Confidentiality of Records	Dec. 81	May 82			
Division 20 Fraud and Suspected Law Violations	Sept. 81	Oet. 81			
Division 21 Civil Rights	Sept. 81	Oct. 81			
*	r			>	χ

ÿ. 8°

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Division 22 State Hearing and Request for Review					
Chapter 000 State Hearing General	Jan. 82	June 82			
Chapter 100 Request for Review	Jan. 82	June 82			•
Chapter 200 Administrative Fraud Hearing	Sept. 81	Oct. 81			
Division 23 Management and Office Procedures					
Chapter 000 Certification for Employer Wage Credit	Sept. 81	Oct. 81			·
Chapter 100 Certification for Employment Wages	Sept. 81	Oct. 81			
Chapter 200 Certification for Employment Wages	Sept. 81	Oct. 81			
Chapter 250 Assignment of State Numbers	Sept. 81	Oct. 81			
Chapter 275 Assignment of State Numbers	Sept. 81	Oct. 81			•
Chapter 300 Assignment of	Sept. 81	Oct. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 350 Disposition of Welfare Records	Sept. 81	Oct. 81			
Chapter 400 Forms Management	Mar. 82	July 82			
Chapter 500 Supplies, Equipment and Forms	Sept. 81	Oct. 81			
Division 24 Building Guidelines	Nov. 81	May 82			
Division 25 Fiscal Management and Control	July 81	Dec. 81			
Division 26 Statistical Reports	Sept. 81	Oct. 81			
Division 27 Advance Contract Pay- ments to Nonprofit Agencies	Sept. 81	Oct. 81			
Division 29 Intra-and Inter- Agency Relations and Agreements	Sept. 81	Oct. 81			
Division 30 Required Service Programs					
Chapter 000 Service Programs Administered by County Welfare Depts	Sept. 81	Oct. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 100 Protective Services for Children	Sept. 81	Oct. 81			
Chapter 150 #3 Protective Services for Adults	Dec. 81	June 82			
Chapter 200 #4 Out-of-Home Care for Children	Dec. 81	June 82			
Chapter 250 #5 Out-of-Home Care for Adults	Dec. 81	June 82			
Chapter 300 #6 Child Day Care Services	Dec. 81	June 82			
Chapter 350 #7 Health	Sept. 81	Dec. 81			
Chapter 400 #8 Family Planning Services	Dec. 81	June 82			
Chapter 450 #9 In-Home Support	Nov. 81	June 82			
Chapter 500 #10 Employment	Sept. 81	Mar. 82			
Division 40 Reception and Application	Dec. 81	June 82		·	
Division 41 Deprivation of Parental Support of Care	Sept. 81	Oct. 81			

ADMINISTRATIVE CODE	PUBLIC COMMENT	STATEMENT OF	METHODS OF	, REVIEW PROCESS	COORDINATION
UNDER REVIEW	ENDING DATE	COMPLETION	. PUBLIC INVOLVEMENT	REVIEW PROCESS	(AOM) INDIAN
Division 42 Nonlinking Factors of Public Assistance	May 82	July 82			
Division 43 Responsible Relatives	Sept. 81	Oct. 81			
Division 44 Standards of Assist- ance	Sept. 81	Oct. 81			
Division 45 Repeal of Special Shelter Payments	Sept. 81	Oct. 81			
Division 46 State Supplemental Program	Sept. 81	Oct. 81			
Division 47 Aid to the Poten- tially Self- Supporting Blind	Sept. 81	Oct. 81			
Division 48 Records	Sept. 81	Oct. 81			
Division 50 Retroactive Benefits	Sept. 81	Oct. 81			
Division 63 Food Stamp Regula- tions	Sept. 81	Oct. 81	·		
Division 67 Low Energy Assistance Program	Sept. 81	Oct. 81			•

APHINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Division 68 Repatriate Americans	Sept. 81	Oct. 81			
Division 69 Refugee Programs	Sept. 81	Oct. 81			

AGENCY: Office of Statewide Health Planning and Development

AGENCY COORDINATOR: Gary Chan

ADDRESS: 714 P Street, Room 1050, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-1212

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
• • • • • • • • • • • • • • • • • • • •	ENDING DATE Aug. 81 May 82 May 81			The hearing for Chapter 1	COORDINATION Chapter 1: 14 Federally Recognize Health System Agencie Chapter 3: Medi-Cal Eligibility Program of the Department of Health Services Chapter 7: Department of Health Services Office of Emergency Services Seismic Safety Commission
Program Chapter 7 Seismic Structural Safety Standards	June 81	Aug. 81			

Page <u>43</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 8 Clinic Renovation (Construction) Grant and Loan	May 82	July 82			
Program					
					•
		·			
				·	

AGENCY: California Unemployment Insurance Appeals Board

AGENCY COORDINATOR: Jo Norton

ADDRESS: 714 P Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-1342

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMEN'T ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 22 Subdivision 2 California Unemployment Insurance Appeals Board Chapter 1 General and Administration Chapter 2 Field Activities Chapter 3	May 81 May 81	Aug. 81 Aug. 81	News Release Mailings to claimants, employer groups, legal aid societies, attorneys and unions Notices will be posted in the California Unemploy- ment Insurance Appeals Board's 11 field offices.	Written comments will be solicited for review by the Board.	Employment Development Department
Appeals Board Activities					

AGENCY: California Occupational Safety and Health Administration

AGENCY COORDINATOR: kon Rinaldi

ADDRESS: 1006 4th Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-3640

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
UNDER REVIEW Title 8 Chapter 3.2 Regulations of the Director and the Division of Industrial Safety Chapter 3.3 Appeals Board Chapter 3.5 Standards Board Chapter 4 Safety Orders Subchapter 1 Unfired Pressure Vessels Subchapter 2 Fired Pressure Vessels Subchapter 3 Compressed Air Subchapter 4 Construction Subchapter 5			PUBLIC INVOLVEMENT	Advisory committees will be chosen by the agency to	Review of Safety and Health regulations will be coordinated with the Division of Occupation Safety and Health. Coordination will also take place with: Building Standards Commission State Fire Marshal State Architect Board/Department of Forestry Department of Housing and Community Develoment Industrial Welfare Commission
Electrical Subchapter 6 Elevator	Sept. 81	Dec. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 6.1 Aerial Tramway	May 81	Oct. 81			
Subchapter 7 General Industry	Nov. 82	Dec. 82			
Subchapter 13 Logging, Sawmill	Nov. 82	Dec. 82			
Subchapter 14 Petroleum Drilling	Oct. 84	Dec. 84			
Subchapter 15 Petroleum Refin- ing	Nov. 84	Dec. 84			
Subchapter 17 Mine	Aug. 81	Dec. 81			
Subchapter 18 Ship and Boat Building	Nov. 84	Dec. 84			
Subchapter 20 Tunne1	Feb. 82	June 82			
Subchapter 21 Telecommunica- tions	Nov. 81	Dec. 81			
				*	,

AGENCY: Department of Industrial Relations

AGENCY COORDINATOR: Dick Gitchell

ADDRESS: 455 Golden Gate Ave., San Francisco, CA 94101

TELEPHONE NUMBER: (415) 557-2810

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 18				ı	
Chapter 8 Rapid Transit Districts	Apr. 81	June 81	Mailing List Trade Journals Press Releases	In the Department of Industrial Relations, each Division will conduct a review of its own regula-	National Labor Relations Board Public Utilities Commission
Chapter 8 Self-Insurance Plans	Apr. 81	June 81	Newspapers Industry Advisory Committees		Local Rapid Transit Districts Department of Insurance
Chapter 8 Division of Labor Statistics and Research - Prevail- ing Wages	May 81	June 81	·	Public meetings for Talent Agencies will be September through March.	
Chapter 7 Division of Labor Statistics and Research - Occupa- tional Injuries	May 81	June 81			Department of Agricul- ture Fair Employment Practic Commission Employment Development
Chapter 2 California Appren- ticeship Council	Oct. 81	Dec. 81		·	Department
Chapter 5 Industrial Homework Act Enforcement	Dec. 81	June 82			
Chapters 5,6,8 Registration of Persons Who Unload	Nov. 81	Feb. 82			
Farm Products					

Page 48

		•			i
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 6 Child Labor Orders	Dec. 81	June 82			
Chapter 6 Security for Wages	Oct. 81	Dec. 81			
Chapter 6 Hearings on Actions to Recover Wages	Oct. 81	Jan. 82			·
Chapter 8 Employment of Aliens Not Entitled to Law- ful Residence	May 81	July 81			
Chapter 8 Uninsured Employers Fund Act Enforcement	Oct. 81	Apr. 82			
Chapter 6 Talent Agencies	Mar. 82	June 82			
			,		
					•

AGENCY: Industrial Welfare Commission

AGENCY COORDINATOR: Margaret Miller

ADDRESS: 455 Golden Gate Ave., San Francisco, CA 94101

TELEPHONE NUMBER: (415) 557-2590

OAL Master Plan For Review of Existing Regulations

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 8 Chapter 5 Division of Labor Standards Enforcement	May 82	June 82	Mailing List which includes Industry Employers Labor Unions Building Trades Councils Wage Board Members Legal Firms	Members of the Wage Board will form an ad hoc task force for review of regulations.	Building Standards Commission Office of Statewide Health Planning and Development (overtime provisions of Public Housekeeping Industry) Department of Education (employment of minors) Department of Rehabilitation (licenses regarding wages and the handicapped) Department of Industrial Relations (internal coordination regarding labor standards enforcement and OSHA)

AGENCY: Department of Boating and Waterways

AGENCY COORDINATOR: Jim Matsueda, Arthur Lee

ADDRESS: 1629 "S" Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5199

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14					
Chapter 1 Department of Boating and Water- ways	July 81	Dec. 81	Mailing Lists Press Releases Public Meetings	Public comments will be reviewed by staff and considered during in-house agency review.	Department of Motor Vehicles Department of Parks and Recreation
Chapter 2 Regulations for Implementation of the California Environmental Quality Act of 1970	July 81	Dec. 81			·
Chapter 3 Conflict of Inter- est Code	July 81	Dec. 81			

AGENCY: California Coastal Commission

AGENCY COORDINATOR: Cynthia K. Long

ADDRESS: 631 Howard Street, San Francisco, CA 94105

TELEPHONE NUMBER: (415) 543-8555

Page <u>51</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14 Division 5.5 Chapter 1 General Provisions Chapter 2 Meetings Chapter 3 Officers and Staff Chapter 5 Coastal Development Permits Chapter 6 Exclusions From Permit Requirement Chapter 7 Coastal Development Permits Issued By Local Government and Other Public Agencies Chapter 8 Implementation Plans Chapter 9 Sewage Treatment Works Chapter 10 Federal Consistency Determination	Mar. 81 Mar. 81 Apr. 81 Apr. 81 June 81	Apr. 81 Apr. 81 Apr. 81 June 81 June 81 July 81 July 81 July 81	Mailing Lists Coastal News Public Meetings	Public written comments will be accepted throughout the review process. Public meetings will be held to accept oral comments. Meetings will be held to accept oral comments as follows: Chapters 1,2,3 February 17-19 San Diego Chapters 5,6 March 3-April 1 San Francisco April 14-16 Marina Del Rey Meetings will complement regular Commission schedules and will be noticed as to specific place at least 10 days prior to the meeting.	Board of Regents of Univeristy of Calif. Board of Trustees of California State University and Colleges Ports of Hueneme, Long Beach, Los Angeles and San Diego Unified Port Districts City of Los Angeles State Water Resources Control Board Office of Coastal Zone Management United States Geologic Survey

AGENCY: State Coastal Conservancy

AGENCY COORDINATOR: Susan Furrer

ADDRESS: 1212 Broadway, Rm. 514, Oakland, CA 94612

TELEPHONE NUMBER: (916) 464-1070

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14 Division 5.6 State Coastal Conservancy	May 81	July 81	Mailing List	Written comments accepted until May 6. Staff recommendations based on public comment to be presented to the Board at its June meeting.	As Required

AGENCY: Colorado River Board

AGENCY COORDINATOR: Dennis Underwood

ADDRESS: 107 South Broadway, Los Angeles, CA 8103

TELEPHONS NUMBER: (213)620-4480

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14					
Division 5.2					·
Chapter 1 Guidelines	Aug. 81	Oct. 81	Mailing List	Background papers will be developed and distributed to mailing list. Public comment will be invited at regularly scheduled monthly Committee meetings.	Resources Agency
	1				

AGENCY COORDINATOR: Joseph J. Henao

AGENCY: Department of Conservation

ADDRESS: 1416 Ninth Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-6369

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14				,	
Division 2			Mailing Lists	The Department will publish	
Chapter 4 Development, Regulation and Conservation of Oil and Gas Resources			Newspapers	notice in newspapers 60 days prior to the due date of public comments.	As required
Subchapter 1 Onshore Well Regulations	Nov. 81	Jan. 82			
Subchapter 1.1 Offshore Well Regulations	Sept. 81	Jan. 82			
Subchapter 2 Environmental Protection	July 81	Jan. 82	·		·
Subchapter 3 Unit Operations	Apr. 81	Jan. 82			
Subchapter 4 State-wide Geo- thermal Regula- tions	May 81	Jan. 82			·
Subchapter 5 Disclosure and Inspection of Public Records		Mar. 81			
Chapter 2 CEQA regulations	Nov. 81	Jan. 82			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
,					
Chapter 8 Mining and Geology					
Subchapter 1 State Mining and Geology Board	July 81	Jan. 82			

AGENCY: Department of Fish and Game

AGENCY COORDINATOR: Harold C. Cribbs

ADDRESS: 1416 9th Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5708

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14					
Part l Fish, Amphibia and Reptiles			Mailing Lists Newspapers Public Meetings	Review process will combine AB 1111 review with Department's separate	The Commission by letter will solicit comments from the Department of
Chapter 1 General Provisions and Definitions	Dec. 81	Dec. 81		statutory review of regulations.	Food and Agriculture during the review process.
Chapter 2 Statewide Regulations for Fishing and Frogging in Inland Waters	Dec. 81	Dec. 81	·		
Chapter 3 Inland Water Fishing	Dec. 81	Dec. 81			
Chapter 4 Ocean Fishing	Dec. 81	Dec. 81			
Chapter 5 Reptiles and Amphibians	Dec. 81	Dec. 81			
Chapter 6 Fish, Commercial	Nov. 81	Nov. 81			
Chapter 7 Capture, Transportation, Possession and Sale of Live Freshwater Fish for Bait for Profit		Jan. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 8 Miscellaneous	Aug. 81	Oct. 81			
Part 2 Game and Furbearers					
Chapter 1 General Provisions and Definitions	Apr. 81	June 81			
Chapter 2 Resident Small Game	Aug. 81	Sept. 81			
Chapter 3 Big Game	Apr. 81	June 81			
Chapter 4 Special Hunts	Apr. 81	June 81			
Chapter 5 Furbearing Mammals	Apr. 81	June 81			
Chapter 6 Nongame Mammals	Apr. 81	June 81			
Chapter 7 Migrating Game Birds	Aug. 81	Sept. 81			
Chapter 8 Public Shooting Grounds	Aug. 81	Sept. 81			
Chapter 9 Shooting Clubs	May 81	June 81			
Chapter 10 Areas Closed to Hunting	Apr. 81	June 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 11 Ecological Reserves	May 81	June 81			
Part 3 General Regulations					
Chapter 1 Collecting Permits	May 81	June 81			
Chapter 3 Miscellaneous	May 81	June 81			
Chapter 4 Regulations for Implementation of the California Environmental Quality Act of 1970	May 81	June 81			
Chapter 5 Fish and Game. Commission, Wild- life Conservation Board, Marine Research Commit- tee, and Depart- ment of Fish and Game-Conflict of Interest Code	Aug. 81	Sept. 81			
Index	Dec. 81	Dec. 81		1	

AGENCY: Department of Forestry and State Board of Forestry

AGENCY COORDINATOR: Deane L. Bennett

ADDRESS: 1416 9th St., 15th Floor, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-9379

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Division 1.5 Chapter 4 Forest Practices Subchapter 1 Abbreviations and Definitions Subchapter 3 Forest District Boundaries Subchpater 2 Application of Forest Practice Rules	Dec. 81	Jan. 82 Dec. 81	Administrative Notice Register Newspapers Mailing List Advisory Committees	A comment period will be provided to allow public to respond to and identify important issues. Public comments received and the Department's and Board's own systematic review of regulations will be used to develop issue papers which will be available upon request.	California Coastal Commission Department of Fish and Game Department of Mines and Geology Department of Parks and Recreation Department of Agriculture Board of Equalization U.S. Forest Service Department of Health Services Water Quality Control Board
Article 1	Oct. 81	Nov. 81			Solid Waste Management
Artiçle 2	Oct. 81	Nov. 81			Board
Article 3	July 81	Aug. 81			
Subchapter 4 Coast Forest District Rules	See Schedu Articles	le for 1-11 below			
Subchapter 5 Northern Forest District Rules	See Schedu Articles	le for 1-11 below	,		·
Subchapter 6 Southern Forest District Rules	See Schedu Articles	le for 1-11 below			•

Page 60

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 1	May 81	June 81			
Article 3	Nov.80	Dec. 80			
Article 4	June 81	July 81			
Article 5	June 81	July 81	1		
Article 6	May 81	June 81			
Article 7	May 81	June 81			
Article 8	May 81	June 81			
Article 9	May 81	June 81			
Article 10	May 81	June 81			
Article 11	May 81	June 81			
Sections 912, 932,952,912.5, 932.5,952.5	June 81	July 81			
Sections 912.7, 932.7,952.7	Dec. 81	Jan. 82			
Subchapter 7 Administration					
Article 1	Aug. 81	Sept. 81			
Article 2	Oct. 81	Nov. 81			
Article 3	July 81	Aug. 81			
Article 4	Aug. 81	Sept. 81			
Article 5	Dec. 81	Jan. 82			
Article 6	Dec. 81	Jan. 82			
Article 7	Aug. 81	Sept. 81			
Chapter 5 Board of Forestry Organization and Operation	·				
Subchapter 1 Board Organiza- tion and Proce- dure	July 81	Aug. 81			
				*	*

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 2 Board Rulemaking Under Forest Practices Act	July 81	Aug. 81			
Subchapter 3 Other Board Actions	July 81	Aug. 81			
Chapter 6 Hazerdous Fire Areas and Condi- tions					
Articles 1,3,4,5,	May 81	June 81			
Chapter 7 Fire Protections					
Articles 1,3 Article 2 Article 4	Sept. 81 May 81 April 81	Oct. 81 July 81 April 81		·	
Chapter 9 State Forests Use and Sales	Aug. 81	Sept. 81			
Subchapter 1 Recreational Use	Aug. 81	Sept. 81			
Subchapter 2 Geothermal Development	Aug. 81	Sept. 81			
Subchapter 3 Timber Sales	Aug. 81	Sept. 81		·	
			1		

Page 62

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 9.5 Forest Improvement Program	Mar. 81	Apr. 81			·
Chapter 9.7 Urban Forestry Program Procedures	Mar. 81	Apr. 81			
Chapter 10 Registration of Professional Foresters	Nov. 81	Dec. 81			· ·
Chapter 11 Policy for Adminis- tration of the Department	July 81	Aug. 81			
Chapter 8 Forest Products Wood Waste Disposal	July 81	Aug. 81			
					·
	A				,

AGENCY: Department of Parks and Recreation

AGENCY COORDINATOR: Andrea Patterson

ADDRESS: 1416 Ninth St, 14th Flr., Sacto, CA 95814

TELEPHONE NUMBER: (916) 445-2358

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
itle 14 Division 3					
Chapter 1 General	June 81	Nov. 81	Mailing Lists Newspapers Public Meetings will be scheduled in May and June	The review will be conducted by selected staff from the Operations, Resource	As required.
Chapter 2 Vehicle, Vehicle Operators, Horses and Bicycles	June 81	Nov. 81	of 1981 in regional head- quarters located in Santa Rosa, Stockton, Monterey, and San Diego. Additional	Protection and Recreational and Local Assistance Divisions, the Office of Off-Highway Vehicle Recreation, and Concessions Office.	
Chapter 3 Concessions	June 81	Nov. 81	meetings will be scheduled if a need is determined.	Public comment will be re- ceived in May and June of	
Chapter 4 Camping	June 81	Nov. 81		1981 following which the Statement of Review Comple- tion will be submitted to	
Chapter 5 Hunting	June 81	Nov. 81		OAL on November 1, 1981.	
Chapter 6 Provisions Limited to Specific Park Units	June 81	Nov. 81			
Chapter 7 Aquatic and Boating	June 81	Nov. 81			
Chapter 8 Winter Sports	June 81	Nov. 81			
Chapter 10 Classification of Park Units	June 81	Nov. 81			

Page <u>64</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 11 Conflict of Interest	June 81	Nov. 81			
Chapter 12 Land and Water Conservation	June 81	Nov. 81			
Chapter 13 Roberti-Z'Berg Urban Open Space and Rec- reation Program	June 81	Nov. 81			
Chapter 14 California Exposi- tion and State Fair	June 81	Nov. 81			
•				·	
			•		
	a).		·	•	->

AGENCY: Reclamation Board

AGENCY COORDINATOR: David B: Anderson

ADDRESS: 1416 9th Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-9873

Page 65_

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 23 Chapter 1 Reclamation Board Chapter 1.5 Plain Management	Mar. 81	May 81 June 81	Mailing Lists Newspapers Public Meeting	Special meetings will be held in conjunction with regular meetings. Hearing Officers will report public input received to the Board.	Department of Water Resources

AGENCY: Resources Agency

AGENCY COORDINATOR: Norman Hill

ADDRESS: 1416 9th Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-9134

The second secon	ROCESS COORDINATION
Title 14 Division 6 Chapter 2 Open Space Subvention Act Chapter 2.5 Dredging Projects in San Francisco Bay Chapter 3 Guidelines for Implementation of the California Environmental Quality Act of 1970 Chapter 4 Resources Agency Conflict of Interest Code May 81 Dec. 81 Dec. 81 Dec. 81 California EIR Monitor (A publication with over 600 subscribers) Mailing Lists Newsletters Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public Meetings Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public commen sion with age ed with the CEnvironmental Public Meetings Public Mee	and discus- cies concern- lifornia Quality Act ll establish changes and Coordination with all entities involved with California Environmenta Quality Act regulations

AGENCY: San Francisco Bay Conservation and Development Commission

AGENCY COORDINATOR: Jonathan Smith

ADDRESS: 30 Van Ness Avenue, San Francisco, CA 94102

TELEPHONE NUMBER: (415) 557-3686

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14					
Division 5			Mailing Lists	Staff reports and recom-	Coordination with other
Chapter l General Provisions	Apr. 81	May 81	Local Newspapers Public Meetings as required ed by expressions of	mendations will be submit- ted to the Commission for action.	local and state agencies will be done in the same manner as solici-
Chapter 2 Commission, Staff and Advisory Boards	Apr. 81	May 81	interest		tation of public comment
Chapter 3 Permit Procedures: General	June 81	Aug. 81			
Chapter 4 Permit Procedures: Commission Hearings	June 81	Aug. 81			
Chapter 5 Permit Procedures: Administrative and Emergency Permits	June 81	Aug. 81			
Chapter 6 Permit Procedures: Permits and Notification of Denial of Application and Certificates of Compliance		Aug. 81		· · · · · · · · · · · · · · · · · · ·	
	·				

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 7 Permit Procedures: Amendments to Permits, Enforcement and Revocation of Permits	May 81	Aug. 81			
Chapter 8 Plan Amendments; Adopting Special Area Plans	July 81	Sept. 81			
Chapter 9 Regulations for Implementation of the California Environmental Quality Act of 1970	Aug. 81	Oct. 81			
Chapter 10 Procedures for Cease and Desist Orders	Apr. 81	June 81			
Chapter 11 San Francisco Bay Conservation and Development Com- mission-Conflict of Interest Code	Apr. 81	May 81			
Chapter 12 Marsh Development Permits	July 81	Sept. 81			
	ę.			,	•

AGENCY: Solid Waste Management Board

AGENCY COORDINATOR: Robert F. Conheim

ADDRESS: 1020 Ninth Street, Suite 300, Sacto, CA 95814

TELEPHONE NUMBER: (916) 322-3330

Page <u>69</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14					
Division 7			Mailing Lists	Issue papers will be used	As required
Chapter 1 General Provisions	June 81	Jan. 82	Trade Publications Advisory Groups will be used in reviewing Chapters	for all chapters and made available to the public. Special emphasis will be	
Chapter 2 Planning Guidelines and Procedures for Preparing, Revising and Amending County Solid Waste Manage- ment Plans		Nov. 81	3,5, and 9	given to the review of Chapters 3,5, and 9.	
Chapter 3 Minimum Standards for Solid Waste Handling and Dis- posal	Apr. 81	Jan. 82			
Chapter 4 Conformance of Solid Waste Facil- ities to County Solid Waste Manage- ment Plans	June 81	Nov. 81			
Chapter 5 Enforcement of Solid Waste Stand- ards and Adminis- tration of Solid Waste Facilities Permits	Apr. 81	Jan. 82			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 7 Litter Control, Recycling, and Resource Recovery Grants	June 81	Jan. 82			
Chapter 8 Used Oil Recycling Program	June 81	Jan. 82			
Chapter 9 Procedures for the Implementation of Solid Waste Disposal Surcharge	Apr. 81	Jan. 82			
				·	
					·

AGENCY: State Water Resources Control Board

AGENCY COORDINATOR: Craig Wilson

ADDRESS: P.O. Box 100, Sacramento, CA 95801

TELEPHONE NUMBER: (916) 322-3580

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 23 Chapter 23 Subchapter 1 General Provisions Subchapter 10 Licensing and Regulation of Use of Oil Spill Cleanup Agents Subchapter 11 Certification Conformance With Water Quality Standards Subchapter 12 Certification Under Section 169 of the Internal Revenue Code of 1954, as Amended Subchapter 18 Areawide Waste Treatment Management Subchapter 19 Certification Under the CA. Pollution Control Financing Authority Act	Mar. 81 Mar. 81 Mar. 81 Mar. 81	May 81 May 81 May 81 May 81 May 81	Mailing Lists Newspapers Trade Journals Advisory Committees Water Quality Coordinating Committee Tri-Agency Technical Advisory Committee Agricultural Advisory Committee Operator Certification Advisory Group California Association of Sanitation Agencies Association of California Water Agencies	The Board plans to conduct workshops and to utilize public meetings and regular Board meetings to facilitate the public participation process. The Board envisions combining the public participation process required for the review of existing regulations with the public hearing process required to make regulation changes.	For the purpose of review, coordination is broken into three levels: Level One: Joint regulations where joint action is required. Level Two: Regulations which may overlap regulations of another agency or which may have a substantial impact on another agency's programs. Level Three: Regulations which impact an agency, but not to a substantial degree. For a complete list of agencies and chapters affected contact the Board.

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 20 Standards for Removal of Sewage From Vessels	Mar. 81	May 81			Coordination efforts will be made with the following agencies: Resources Agency
Subchapter 20.1 Designation of Marine Terminals to Provide Adequate Sewage Retention Beriu Pumpout Facilities	Mar. 81	May 81			Department of Water Resources Department of Food and Agriculture Department of Health Services Department of Fish and Game
Subchapter 21 Certification for Small Business Concerns Applying for Small Business Act Loans to Meet Water Pollution Control Requirements	Mar. 81	May 81			and Game Division of Forestry
Subchapter 5 Loans to Public Agencies	Apr. 81	June 81			
Subchapter 7 Clean Water Grant Program	Apr. 81	June 81			
Subchapter 8 Other Federal Grant Programs	Apr. 81	June 81			
Subchapter 14 Classification of Waste Water Treat- ment Plants and Operator Certifica- tion	Apr. 81	June 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 2 Appropriation of Water	May 81	July 81			
Subchapter 2.5 Water Rights for Stockponds	May 81	July 81			
Subchapter 3 Determination of Rights to the Use of Water	May 81	July 81			
Subchapter 4 Recordation of Water Extractions and Diversions	May 81	July 81			
Title 23			er en		
Chapter 5 Prevention of Waste, Unreasonable Use or Diversion of Water	May 81	July 81			
Subchapter 6 Review by State Board of Action or Failure to Act by Regional Board	June 81	Aug. 81			
Subchapter 9 Waste Discharge Reports and Re- quirements	June 81	Aug. 81			
Subchapter 9.1 Enforcement Pro- cedures and Septic	June 81	Aug. 81			
Tank Prohibition Review by Board		* diplomatical section of the sectio			

Page <u>74</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 15 Waste Disposal to Land	June 81	Aug. 81			·
Title 23 Chapter 4 Regional Water Quality Control Board	June 81	Aug. 81			
,					Þ

Page 75

AGENCY: Department of Water Resources

AGENCY COORDINATOR: John Kramer

ADDRESS: 1416 Ninth Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-9951

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 1 Chapter 2 Subchapter 3 Arbitration of Disputes Under State Contract Act Contracts Title 23	May 81	June 81	Title 23, Chapter 1.5 Reclamation Board mailing list, publication in news- papers in Cobey-Alquist floodplain management areas Chapter 2, Subchapter 1 Notice to be sent to each dam owner. Chapter 2, Subchapter 1.5	Water Commission. These will be followed by staff review and will subsequent-ly be brought back to the	Services. Title 23, Chapter 1.5 Joint Board and Department review is planned.
Chapter 1.5 Flood Plain Management Chapter 2 Department of Water Resources	Mar. 81	June 81	Newspapers in areas which have levees controlled by the Department. Chapter 2, Subchapter 1.6 Newspapers, notice in Sacramento Bee, March 12, 1982. Chapter 2, Subchapter 1.7	Commission for further consideration. In some cases workshops are scheduled. Most meetings are planned for Sacramento.	Chapter 2 U.S. Army Corps of Engineers Chapter 2, Subchapter 1.6 Housing and Community Development General Services
Subchapter l Dams and Resevoirs	May 81	June 81	Architectural and Engineer- ing Professional Societies Notices in newspapers in		Chapter 2, Subchapter 1.7 Department of Trans-
Subchapter 1.5 Levee and Right of Way Control	Apr. 81	June 81	Los Angeles and Sacramento Chapter 2, Subchapter 2 Sacramento Bee Chapter 2, Subchapter 2.5		portation Department of General Services Chapter 2, Subchapter
Subchapter 1.6 Relocation Assis- tance Program	May 82	July 82	Newspapers Chapter 2, Subchapter 4 Mailing List		4 Department of Fish and Game
Subchapter 1.7 Process for Private Architectural and Engineering Firms	Mar. 81	June 81	Newspapers Board of Supervisors in Project Counties		Department of Food and Agriculture State Lands Commission

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 2 Financial Assis- tance for Local Projects	Apr. 81	June 81			California Regional Planning Agency California Regional Water Quality
Subchapter 2.5 California Safe Drinking Water Bond Law of 1976	Apr. 81	June 81			Control Board
Subchapter 2.6 Weather Resources Management	Apr. 81	June 81			
Subchapter 3 Implementation of California En- vironmental Qual- ity Act	Apr. 81	June 81			
Subchapter 4 Conflict of Interest	Apr. 81	June 81			·
Chapter 5					, miles.
Subchapter 1 Prevention of Waste, Unreason- able Use or Diversion of Water	Apr. 81	June 81			
·	·				
				,	

AGENCY: Board of Accountancy

AGENCY COORDINATOR: Della Bousquet

ADDRESS: 2135 Butano Dr., Suite 112, Sacto, CA 95825

TELEPHONE NUMBER: (916) 920-7121

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 1		0.5 01	M. M		
State Board of Accountancy	Sep. 81	Oct. 81	Mailing List Trade Journals Newspapers Advisory Committees	After the review process, regulatory changes will be considered at a public meeting in January 1982.	Department of Consumer Affairs
	·				

AGENCY: Office of Administrative Hearings

AGENCY COORDINATOR: Herb Nobriga

ADDRESS: 717 K Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4926

Page <u>78</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 1 Chapter 2 Contract Hearing Procedures	Feb. 82	Mar. 82	Newspapers Industry Publications (AGC Monthly) Arbitration Committee	Legislation has been introduced (AB 1227) which will substantially modify the function of OAH. It will result in the repeal of Subchapter 2 and substantial change to Subchapter 3. Following enactment of the legislation, regulatory proceedings will occur.	Arbitration Committee composed of 7 members; representatives of the construction industry, three major contracting state agencies (Genera Services, Transportation, and Water Resources) and Office of Administrative Hearings.

AGENCY: California State Board of Architectural Examiners

AGENCY COORDINATOR: Paul Welch, Jr.

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-3393

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16		:			
Chapter 2 California State Board of Architectural Examiners	Jan. 82	Mar. 82	Mailing List Department of Consumer Affairs Mailing List Newspapers	Following completion of the review process, regulatory changes will be considered at two hearings scheduled for April and June of 1982.	Department of Consumer Affairs Office of the State Architect Department of Rehabilitation Department of Housing and Community Development
	,				

AGENCY: State Athletic Commission

AGENCY COORDINATOR: Steve English

ADDRESS: 1021 "0" Street, Rm. A-153, Sacto, CA 95814

TELEPHONE NUMBER: (916) 445-7898

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 2			Mailing List	Subchapters 1 and 1.5 will	Department of Consumer
Subchapter 1 Professional Boxing	Aug. 81	Dec. 81	Trade Journal	be open to public com- ment in July 1981. Sub- chapters 2 and 3 will be open to public comment	Affairs
Subchapter 1.5 Karate	Aug. 81	Dec. 81		in October 1981.	
Subchapter 2 Amateur Boxing	Nov. 81	Dec. 81			
Subchapter 3 Wrestling	Nov. 81	Dec. 81			

AGENCY: Bureau of Automotive Repair

AGENCY COORDINATOR: Wanza Sizemore

ADDRESS: 3116 Bradshaw Road, Sacramento, CA 95827

TELEPHONE NUMBER: (916) 366-5028

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC STATEM COMMENT OF ENDING DATE COMPLE	METHODS OF	REVIEW PROCESS	COORDINATION
Title 16 Chapter 33 Bureau of Automotive Repair Subchapter 1 Automotive Repair Dealers and Official Stations Adjusters, and Installers Subchapter 2 Motor Vehicle Inspection Program	Dec. 81 Dec. 8		After the review process, regulatory changes will be considered at a public meeting to be scheduled. Public meetings are planned for June, August, October and December. A Statement of Completion is to be forwarded to the Office of Administrative Law by the end of December.	Department of Consumer Affairs

AGENCY: State Board of Barber Examiners

AGENCY COORDINATOR: James Knauss

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-7008

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16					
Chapter 3 State Board of Barber Examiners	Sept. 81	Oct. 81	Trade publications Board bulletins Newspapers Consumer Groups	Following completion of the review process, regulatory changes will be considered at hearings scheduled for:	State Building Stand- ards Commission Board of Cosmetology
				March 1982 in San Francisco	
				May 1982 in Los Angeles	
	·				

AGENCY COORDINATOR: Samuel Levin

AGENCY: Board of Behavioral Science Examiners

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4933

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 18 Board of Behavioral Science Examiners	Feb. 82	Mar. 82	Direct mail to licensees, registrants and applicants Special mailing list, supplied by Department of Consumer Affairs, including organizations and associations with interest in consumer rights. Newspapers Legislative Journals Advisory Committees	Regulatory changes will be considered at Board meetings to be held: May 1981 November 1981 March 1982 April 1982	Department of Consumer Affairs

AGENCY: Cemetery Board

AGENCY COORDINATOR: John Gill

ADDRESS: 1430 Howe Avenue, Suite 88, Sacto, CA 95825

TELEPHONE NUMBER: (916) 920-6078

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 23 State Cemetery Board		Sept. 81	Newspapers Trade journals Mailing list Task Force	After completion of review process, regulatory changes will be considered by the Board at a meeting tentatively scheduled for October in Southern California	Board of Funeral Directors and Embalmers
		·			

AGENCY: Certified Shorthand Reporters Board

AGENCY COORDINATOR: Judy Tafoya

ADDRESS: 1021 O Street, Room A-141, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5101

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 24 Certified Shorthand Reporters Board	Mar. 82	Mar. 82	Mailing List Notices in Professional Journals	Following completion of the review process, regulatory changes will be considered by administrative staff.	State Bar Municipal, Superior, Appellate Courts (Clerks and Bench) California Judicial Council Western Center on Law and Poverty Legal Services Corporation California Rural Legal Assistance Secretary of State Office of Private Postsecondary Education Community Colleges (Chancellor's Office) Senate and Assembly Committees

AGENCY: Bureau of Collection and Investigative Services

AGENCY COORDINATOR: Cindy Thompson

ADDRESS: 1430 Howe Avenue, Sacramento, CA 95825

TELEPHONE NUMBER: (916) 920-6435

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16					
Chapter 7 Bureau of Collection and Investigative Services			Licensees, Police and Sheriff's departments and	After the conclusion of the review process, regulatory changes will be considered at hearings to be scheduled.	Department of Consumer Affairs
Article l General Provi- sions	June 81	Aug. 81	associations, banking and lending associations		
Article 2 Collection Agencies	June 81	Aug. 81			
Article 3 Private Investigators, Private Patrol Operators, Insurance Adjustors, Repossessors and Alarm Company Operators	June 81	Aug. 81			
	·				

AGENCY: Board of Registered Construction Inspectors

AGENCY COORDINATOR: Lisa Frederiksen

ADDRESS: 1434 Howe Avenue, Sacramento, CA 95825

TELEPHONE NUMBER: (916) 323-2556

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 34 Board of Registered Construction Inspectors	June 81	Sept. 81	Mailing List Newspapers Advisory Committees	Following completion of review process, regulatory changes will be considered by the Board.	Office of the State Architect
·	·		·		
	·				
	-				
			·		

AGENCY: Contractor's State License Board

AGENCY COORDINATOR: Glenn Hair

ADDRESS: 3132 Bradshaw Road, Sacramento, CA 95826

TELEPHONE NUMBER: (916) 366-5263

Title 16 Chapter 8 Article 2 Application for License Article 3 Classification Sections 755-760 Article 4 Examinations Article 3 Classification Sections 730-754 Article 1 General Policies Articles 5-7 Renewals, Boards and Special Provisions Articles 5-7 Renewals, Boards and Special Provisions Article 10 Apr. 82 June 82 Mailing List Newspapers Notice Register Mailing List Newspapers Trade Journals Notice Register Issue papers will be available upon request. June 82 June 82 June 82 June 82 Apr. 82 June 82 June 82 Apr. 82 June 82	ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 2 Application for License Article 3 Classification Sections 755-760 Article 4 Examinations Article 3 Classification Sections 730-754 Article 1 General Policies Article 5-7 Renewals, Boards and Special Affairs Newspapers Trade Journals Notice Register All June 82 June 82 June 82 June 82 App. 82 Ap	Title 16					
Article 2 Application for License Article 3 Classification Sections 755-760 Article 4 Examinations Article 3 Classification Sections 730-754 Article 1 General Policies Articles 5-7 Renewals, Boards and Special Trade Journals Notice Register Trade Journals Notice Register Trade Journals Notice Register Trade Journals Notice Register	Chapter 8				Issue papers will be avail-	Department of Consumer
Classification Sections 755-760 Article 4 Oct. 81 June 82 Examinations Article 3 Jan. 82 June 82 Classification Sections 730-754 Article 1 Apr. 82 June 82 General Policies Articles 5-7 Apr. 82 June 82 Renewals, Boards and Special	Application for	Oct. 81	June 82	Trade Journals	able upon request.	Affairs
Article 4 Examinations Article 3 Classification Sections 730-754 Article 1 General Policies Articles 5-7 Renewals, Boards and Special Oct. 81 June 82 June 82 June 82 June 82 Apr. 82 June 82 June 82		Oct. 81	June 82			·
Examinations Article 3 Classification Sections 730-754 Article 1 General Policies Articles 5-7 Renewals, Boards and Special June 82 June 82 June 82 June 82 June 82	Sections 755-760					
Classification Sections 730-754 Article 1 General Policies Articles 5-7 Renewals, Boards and Special Apr. 82 June 82 June 82		Oct. 81	June 82			
Article 1 General Policies Articles 5-7 Renewals, Boards and Special Apr. 82 June 82 June 82		Jan. 82	June 82			
General Policies Articles 5-7 Apr. 82 June 82 Renewals, Boards and Special	Sections 730-754					
Renewals, Boards and Special		Apr. 82	June 82			
	Renewals, Boards and Special	Apr. 82	June 82			
			·			
						·
						·
						·
				,		

AGENCY: Board of Cosmetology

AGENCY COORDINATOR: Delphine Cathcart

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-9278

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 9 Board of Cosmetology Articles 1,3,8 Article 2 Articles 4,6 Article 7	Mar. 81 July 81 Nov. 81 Mar. 82	Apr. 81 Aug. 81 Dec. 81 Apr. 82	Trade Journals Mailing List	Following completion of the review process, regulatory changes will be considered at hearings scheduled for May, September of 1981 and January, May of 1982.	State Board of Barber Examiners Department of Health Services State Building Stand- ards Commission

Page 90

AGENCY: Board of Dental Examiners

AGENCY COORDINATOR: Celia Samford

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-6197

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 10 Board of Dental Examiners Articles 3,4,6,7, 8,9,10,13 Articles 1,2,5, 8.5,11,12,14	Oct. 81	Nov. 81 May 82	Trade Journals Consumer Publications Mailing List Board Committees Newspapers	After the review process, regulatory changes will be considered by the Board. Staff summaries of public input will be prepared and made available to the public.	Department of Consumer Affairs

AGENCY: Bureau of Electronic and Appliance Repair

AGENCY COORDINATOR: William Hayes

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4751

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16					
Chapter 27 Bureau of Electronic and Appliance Repair		Jan. 82	Department of Consumer Affairs' Information Office press release Mailing List	Following completion of the review process, regulatory changes will be considered at hearings scheduled for	Department of Consumer Affairs
			Newsletter Advisory Board	September 1981 in Los Angeles	
				March 1982 in Sacramento	
	·				

AGENCY: Bureau of Employment Agencies/Nurses' Registry

AGENCY COORDINATOR: Portia Siplin

ADDRESS: 1333 Howe Avenue, Sacramento, CA 95825

TELEPHONE NUMBER: (916) 920-6311

Page <u>92</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 28 Bureau of Employment Agencies Subchapter 1 Subchapter 2	Oct. 81 Sept. 81	Dec. 81	Newspapers Mailing List Advisory Board	Following completion of review process, regulatory changes will be considered at hearings to be scheduled.	Department of Consumer Affairs
	·				
			·		

AGENCY COORDINATOR: william Peterson

AGENCY: State Board of Fabric Care

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-7686

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16					
Chapter 6 State Board of Fabric Care	Apr. 82	May 82	Trade Publications Newspapers Mailing List Consumer Groups Task Force made up of five- industry members, two- general consumer public, two-other arms of govern-	Task Force will review all regulations, staff comments, public and industry comments and make recommendations to Board at regularly scheduled regulation hearings.	State Department of Health State Fire Marshal CAL-OSHA
			ment.		,
				_	
				-	

AGENCY: Fair Employment and Housing Commission

AGENCY COORDINATOR: David A. Garcia A.

ADDRESS: 30 Van Ness Ave., 3rd Floor, San Francisco, CA 94102

TELEPHONE NUMBER: (415) 557-2325

Page 94_

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 4 Chapter 1 General Matters Chapter 2 Particular Employment Practices Chapter 4 National Origin Discrimination	Sept. 81 Oct. 81 May 81	Dec. 81 Dec. 81 July 81	Newspapers Trade Journals Notice Register Mailing Lists Public Meetings	Public meetings will be scheduled in July or August 1981, in Los Angeles or San Francisco.	State and Consumer Services Agency State Personnel Board Health and Welfare Agency Employment Development Department State Agencies as required
Chapter 5 Ancestry Discrimination	May 81	July 81			
	,				
		ı	;		-

Page 95

AGENCY: State Fire Marshal

AGENCY COORDINATOR: Mr. Pat McLaughlin

ADDRESS: 7171 Bowling Dr., Ste. 800, Sacto, CA 95823

TELEPHONE NUMBER: (916) 322-2370

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 1 Subchapter 1 General Construction Subchapter 2 Tents Subchapter 4 Fire Alarms Subchapter 7 Wearing Apparel Subchapter 8 Flame Retardant Chemicals Subchapter 9 Fire Equipment Threads Subchapter 3 Fire Extinguishers Subchapter 6 Fireworks Subchapter 10 Explosives Subchapter 11 Transportation of Flammable Liquids	May 81 May 81 May 81 May 81 May 81 May 81 Nov. 81 Nov. 81 Nov. 81	June 81 June 81 June 81 June 81 June 81 June 81 June 82 Jan. 82 Jan. 82 Jan. 82	Notices in General Circulation Newspapers in Los Angeles, San Diego, San Francisco, Fresno, and Sacramento. Notice in California Notice Register: Phase 1 4/1/81 (Subchapters 1, 2, 4, 7, 8 and 9) Phase 2 9/18/81 (Subchapters 3, 6, 10, 11, 11.5 and 12)	Members of the public desiring to submit comments should do so by: May 20, 1981 for Subchapters 1,2,4,7,8,9 and November 4, 1981 for Subchapters 3,6,10,11,11.5, 12. Meetings are scheduled in Los Angeles on May 19, 1981 and November 3, 1981 and in Sacramento on May 20, 1981 November 4, 1981	Housing and Community Development State Architect Statewide Health Planning and Development Energy Resources Conservation and Development Commission Division of Fairs and Expositions General Services Highway Patrol Division of Industrial Safety Air Resources Board Secretary of State Building Owners and Managers Association California Building Officials International Conference of Building Officials Gas Producers Fire Equipment Manufacturers' Association Fireworks Importers/ Exporters Fireworks Wholesalers Department of Health Services

	PUBLIC	STATEMENT	·		
ADMINISTRATIVE CODE UNDER REVIEW	COMMENT	OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 11.5 Gasoline Vapor Control Systems	Nov. 81	Jan. 82			Advisory Committees: Fireworks Fire Alarm
Control Systems Subchapter 12 Portable Internal Combustion Engine- Driver Pumps	Nov. 81	Jan. 82			Fire Alarm Flammable Liquids Testing, Organization and Inspection Service Agency Fire Suppression California Fire Service Local Government Fire Departments California Fire Chiefs' Association North and South Fire Prevention Officers Section
				·	

AGENCY: Franchise Tax Board

AGENCY COORDINATOR: Paul Petrozzi

ADDRESS: Aerojet Center, Sacramento, CA 95857

TELEPHONE NUMBER: (916) 355-0994

OAL Master Plan For Review of Existing Regulations

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC ST COMMENT ENDING DATE CO	TATEMENT OF OMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 18 Subchapter 1 General Administration Subchapter 2.5 Personal Income Tax Regulations Group 1 General Provisions and Definitions Group 2 Imposition of Tax Group 3 Computation of Taxable Income Group 4 Corporate Distributions and Adjustments Group 5 Deferred Compensation Group 6 Accounting Periods and Methods of Accounting	Apr. 81 Feb. 81 Feb. 81 Mar. 81	Apr. 81 May 81 May 81 May 81 May 81 May 81	Mailing Lists Legal Reporter Services and Clearinghouses CPA Society Tax Section of the State Bar	Internal task forces will be used for preliminary review. Two oral testimony meetings are scheduled for April and August 1981. Other public input dates shown are for submission of written comments; retroactive written comment will be allowed. Entire schedule to be mailed with notice to facilitate written comment. Tax Board will meet to discuss the review in addition to the scheduled public meetings. Additional public meetings will be scheduled as necessary.	Employment Development Department - Wage Withholding Energy Commission - Tax Credits Board of Equalization- Appeals Controller - Senior Citizen Property Tax Postponement Department of Finance- Fiscal Impact to State Multistate Tax Commis- sion - (§§25120- 25139) Attorney General-Tax Law Counsel Interagency Tax Collec- tion Commission - Offset Program

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Group 7 Exempt Trusts and Common Trust Funds	Apr. 81	May 81			
Group 8 Natural Resources	Apr. 81	May 81			
Group 9 Estates, Trusts, Beneficiaries and Decedents	Apr. 81	May 81			
Group 10 Partners and Partnerships	May 81	Aug. 81			
Group 11 Gross Income of Nonresidents	May 81	Aug. 81			
Group 12 Credit for Taxes Paid	May 81	Aug. 81		·	
Group 13 Gain or Loss on Disposition of Property	May 81	Aug. 81			
Group 14 Capital Gains and Losses	June 81	Aug. 81			
Group 15 Readjustment of Tax Between Years and Special Limitations	June 81	Aug. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Group 17 Returns	June 81	Aug. 81			
Group 18 Payments and Assessments	July 81	May 81			
Group 19 Collection of Tax	July 81	May 81			
Group 20 Overpayments and Refunds	July 81	May 81			
Group 21 Administration of Tax	July 81	May 81			
Subchapter 3.5 Bank and Corporate Tax Regulations	Aug. 81				·
Group 1 General Provisions and Definitions	Aug. 81	Nov. 81			
Group 2 The Bank and Corporation Franchise Tax	Aug. 81	Nov. 81			·
Group 3 Corporation In- come Tax	Aug. 81	Nov. 81			
Group 4 Exempt Corpor- ations	Sept. 81	Nov. 81			
Group 5 Computation of	Sept. 81	Nov. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Group 6 Gross Income	Sept. 81	Nov. 81			
Group 7 Net Income	Oct. 81	Nov. 81			
Group 8 Corporate Distri- butions and Ad- justments	Oct. 81	Nov. 81			
Group 9 Corporate Liqui- dations	Oct. 81	Nov. 81			
Group 10 Corporate Organ- izations and Reorganizations	Nov. 81	Feb. 82	·		
Group 11 Insolvency Reor- ganizations	Nov. 81	Feb. 82			
Group 12 Pension Plan	Nov. 81	Feb. 82			
Group 13 Accounting Periods and Methods of Accounting	Dec. 81	Feb. 82			
Group 14 Natural Resources	Dec. 81	Feb. 82			
Group 15 Gain or Loss on Disposition of Property	Dec. 81	Feb. 82	·		

Page 101

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Group 16 War Loss Recover- ies	Jan. 82	Feb. 82			
Group 17 Allocation of Income	Jan. 82	Feb. 82			*
Group 18 Renegotiations	Jan. 82	Feb. 82			·
Group 19 Returns and Pay- ments	Feb. 82	Apr. 82			
Group 20 Assessments	Feb. 82	Apr. 82	-		·
Group 21 Interest, Addi- tions to Tax and Penal Provisions	Feb. 82	Apr. 82			
Group 22 Refunds and Credits	Mar. 82	Apr. 82			·
Group 23 Collection of Tax	Mar. 82	Apr. 82			
•					
	-				

AGENCY: Board of Funeral Directors and Embalmers

AGENCY COORDINATOR: Kathleen Callanan

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-2413

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16					S
Chapter 12 Board of Funeral Directors and Embalmers	Sept. 81	Nov. 81	Press Release through Department of Consumer Affairs Mailing List Trade Magazines Consumer Organization Advisory Committees	Board staff will prepare issue papers on all regulations which will be available at informational hearings and upon written request.	Cemetery Board Department of Consumer Affairs
			providing input on three specific issues:		
			1. licensing of funeral directors 2. licensing of embalmers		
•			3. preneed funeral arrangements		

AGENCY: Department of General Services

AGENCY COORDINATOR: Marjorie Holt

ADDRESS: 915 Capitol Mall, Ste. 590, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4084

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 21 - Public Works Chapter 1-Dept. of General Services Subchapter 1 Office of State Architect Group 1 Safety of Construction of Public Schools Group 2 Access to Public Buildings By Physically Handicapped Persons Group 3 Design Criteria, Materials, and Details of Construction Group 4 Selection Process for Private Architectural	May 81 May 81 May 81		PUBLIC INVOLVEMENT Title 21, Chapter 1 - Public comments will be solicited through newspaper publications in	Title 21, Chapter 1 - Groups 1,2,3: Background papers will be mailed to those on mailing lists and made available to public at Sacramento, San Francisco, and Los Angeles offices. Public meetings are scheduled in the same places in mid-May 1981. Group 4: Will develop background papers prior to soliciting comments. Will have meetings in Sacramento, and in Los Angeles and San Francisco if required. Title 2, Subchapter 2(Art. 1): Filing of regulations expected March 1981, at which time regulations will be reviewed by OAL. Subchapter 4 (Group 1):	COORDINATION Title 21, Chapter 1 State Advisory Board of the Office of State Architect for the Field Act State Superintendent of Public Instruction Public School Associations State Fire Marshal General Services Small Business Office California Office of Minority Business Enterprises Title 2, Subchapter 2 (Article 1): American Civil Liberties Union Peace Officers Research Association of California International Society of Krishna Consciousness Department of Motor

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Rehabilitation and Replacement; School Sites-Nonuse Sections 1863.1-1864.10 Lease-Purchase Law of 1976; State School Deferred Maintenance	Oct. 81	Nov. 81	Subchapter 8: Notices will be published on April 15, 1981 in Los Angeles Times, San Francisco Examiner, and Sacramento Bee.		Fish and Game Department of Trans- portation Subchapter 7: General Services Office of Small Business Legal Office Board of Control Small Business Expansion Council
Sections 1865.1- 1866.15	Oct. 81	Dec. 81	,		Subchapter 8: General Services Office of Procurement
Subchapter 6 Real Estate Re- location Assistance Program	Oct. 81	Dec. 81			Architect Legal Office Small Business Expansion Council
Subchapter 7 Office of Procure- ment	June 81	Aug. 81			Board of Control
Subchapter 8 Office of Small Business Procure- ment and Contracts	June 81	Aug. 81			
			,		

AGENCY: Board of Registration for Geologists and Geophysicists

AGENCY COORDINATOR: John Wolfe

ADDRESS: 1020 N Street, Room 421, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-1920

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 29 Board of Registration for Geologists and Geophysicists	Nov. 81	Feb. 82	Mailing List Professional Journal Advisory Committee	Staff summaries of information submitted during public comment period will be presented at a hearing to be scheduled.	Department of Consumer Affairs

AGENCY: State Board of Guide Dogs for the Blind

AGENCY COORDINATOR: Manuel Urena

ADDRESS: 830 K Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-9040

ADMINISTRATIVE CODE UNDER REVIEW	COM	BLIC MENT NG DATE	(EMENT OF LETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COOR	RDINATION
Citle 16 Chapter 22 Article 1 General Provisions Article 2 Licenses and Certificates of Registration	Dec.		Mar.		Newspapers Braille and taped material accessible to blind per- sons Announcements at conven- tions and meetings of blind persons Mailing Lists	Following the review period, amendments or proposed changes to regulations will be considered at Board meetings tentatively scheduled for April 15, 1982 in Los Angeles and May 22, 1982 in Sacramento.	Affairs	of Consumer
Article 3 Training Schools	Dec.	81	Mar.	82	Mailing Lists	Sacramento.		
Article 4 Standards of Train- ing and Instruction		81	Mar.	82				
Article 5 Miscellaneous	Dec.	81	Mar.	82				

AGENCY: Bureau of Home Furnishings

AGENCY COORDINATOR: Gordon Damant

ADDRESS: 3485 Orange Grove Ave., North Highlands, CA 95660

TELEPHONE NUMBER: (916) 920-6951

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Fitle 4 Chapter 3 Bureau of Home Furnishings	June 81	Aug. 81	Mailing List Consumer Groups Advisory Board	Following completion of the review process, regulatory changes will be considered at three hearings to be scheduled in mid-1981.	Department of Consumer Affairs
• •					
					•
			·		
			·		:

AGENCY: California Horse Racing Board

AGENCY COORDINATOR: Leonard Foote

ADDRESS: 455 Capitol Mall, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-9228

Page <u>109</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 4			Mailing List	Regular Board meetings will	Department of Industria
Article 1 Racing Board Power and Jurisdiction Article 2	June 81	Nov. 81	Horseman's Journal and Daily Racing Form Advisory Committee of appointees from racing, horsemen, breeding	be used for public comment- ing during the review process.	Relations Department of Insurance Department of Food and Agriculture Board of Examiners in
Definitions Article 3	June 81	Nov. 81	associations and labor unions.		Veterinary Medicine
Racing Associations Article 4 Occupational Licenses	June 81	Nov. 81			
Article 5 Racing Officials	June 81	Nov. 81			·
Article 6 Entries and Declarations	July 81	Nov. 81			
Article 7 Claiming Races	July 81	Nov. 81			
Article 8 Running the Race	July 81	Nov. 81	·		
Article 9 Harness Racing Rules	July 81	Nov. 81			
Article 10 Quarter Horse Racing Rules	July 81	Nov. 81			
Article 11 Objections and Protests; Appeals	July 81	Nov. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 12 Colors, Stable	Aug. 81	Nov. 81			
Names, Agents Article 13 Bloodstock Agents	Aug. 81	Nov. 81			
Article 13.5 Authorized Horse	Aug. 81	Nov. 81			
Sales Article 14 California-Bred Awards	Aug. 81	Nov. 81			
Awards Article 15 Veterinary Practices	Sept. 81	Nov. 81			
Article 16 General Conduct	Sept. 81	Nov. 81			
Article 17 Fire Prevention and Security	Sept. 81	Nov. 81			
Article 18 Parimutuel Wagering	Oct. 81	Nov. 81			
Article 19 Policing the Public Inclosure	Oct. 81	Nov. 81			
Article 21 The Standardbred Sires Stakes Program	Aug. 81	Nov. 81			
Article 22 Horsemen's Organ-	Aug. 81	Nov. 81			
izations and Agreements Index					

AGENCY: Board of Landscape Architects

AGENCY COORDINATOR: Joe Heath

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4954

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 26 California State Board of Landscape Architects	Sept. 81	Nov. 81	Newspapers Trade Journals Newsletters	After completion of review process, regulatory changes will be considered at Board meetings to be scheduled.	Department of Consumer Affairs

AGENCY: Board of Medical Quality Assurance

AGENCY COORDINATOR: Gregory Gorges

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-4216

Page <u>112</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Citle 16					
Chapter 13 Board of Medical Quality Assurance	Nov. 81	Jan. 82	Newsletter Mailing List Trade Journals	Issue papers prepared by staff, consultants, or task forces and testimony	Department of Health Services Board of Registered
Chapter 13.1 Psychology Examining Committee	Nov. 81	Jan.82	Consumer and Public Inter- est Groups Task Force Special Subcommittees	presented at hearings will be used to determine regu- latory changes at formal regulation meetings open	Nursing Board of Behavioral Science Examiners
Chapter 13.2 Physical Therapy Examining Committee	Nov. 81	Jan. 82	Special Consultants	to the public.	
Chapter 13.3 Hearing Aid Dispensers Examining Committe	Nov. 81	Jan. 82			
Chapter 13.4 Speech Pathology and Audiology Examining Committee	Nov. 81	Jan. 82			·
Chapter 13.5 Registered Dispensing Opticians	Nov. 81	Jan. 82			
Chapter 13.6 Acupuncture Advisory Committee	Nov. 81	Jan. 82			
Chapter 13.7 Physician's Assistants Examining Committee	Nov. 81	Jan. 82			

φφ						Page 113
9—82195 9—82221	WINITIATOTEMITTAE CODE	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF . PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
	Chapter 13.8 Podiatry Examining Committee	Nov. 81	Jan. 82			
	•					
	·					
						·
					,	÷
		*				
			ē			

AGENCY: California Museum of Science and Industry

AGENCY COORDINATOR: Mr. Chris Dimen

ADDRESS: 700 State Drive, Exposition Park, Los Angeles, CA 90037

TELEPHONE NUMBER: (213) 744-7472

Page <u>114</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2					
Division 3			Written comments to be	Notices will be posted on	Office of Attorney
Chapter 3			solicited through posted notices and newspaper	public boards in lobbies of Main Museum and Space	General Los Angeles Police
Article 1 General	May 81	May 81	(L.A. Times) announcements	Museum. Information also available	Department
Article 2 Traffic		·		from Museum's Public Information Officer.	
Sections 5000- 5009	May 81	May 81	·		·
Sections 5010- 5039	July 81	July 81			
Sections 5040- 5059	Sept. 81	Sept. 81			
Sections 5060- 5066	Nov. 81	Nov. 81			
Article 3 Admission Ticket Selling	Nov. 81	Nov. 81			
			,		

AGENCY: Board of Registered Nursing

AGENCY COORDINATOR: Ken Gray

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-8603

Page <u>115</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 14	Sept. 81	Nov. 81	Professional Journals Newsletters	Regulatory changes to be considered by the Board and	Department of Health Care Services
Board of Registered Nursing		·	Newsletters Mailing List	presented at public hearings	
					Board of Medical Quality Assurance
				•	·

AGENCY: State Board of Examiners of Nursing Home Administrators

AGENCY COORDINATOR: Hal Tindall

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-8435

Page <u>116</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 31 State Board of Examiners of Nursing Home Administrators	Feb. 82	March 82	Mailing List Agency Newsletters	After completion of review process, regulatory changes will be considered at regularly scheduled Board meetings to be held in:	Department of Consumer Affairs
			<i>y</i>	Sacramento June 1981	•
				Los Angeles December 1981 April 1982	
	·				·

Page 117

ADDRESS: 1021 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-2095

AGENCY: Board of Optometry

AGENCY COORDINATOR: John T. Quinn

		T			·
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16					
Chapter 15 Board of Optometry	Oct. 81	Dec. 81	Mailing List Consumer Groups	Following completion of review process, regulatory changes will be considered at Board meetings scheduled for: January 1982 in San Francisco	
				April 1982 in Los Angeles	·

AGENCY: State Personnel Board

AGENCY COORDINATOR: Richard Sanchez

ADDRESS: 801 Capitol Mall, Sacramento, CA 95814

TELEPHONE NUMBER: __(916) 322-2770

OAL Master Plan For Review of Existing Regulations

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division I Administrative Personnel Chapter 1 State Personnel Board			Notice Register Memorandum to all state agencies and employee organizations Newspapers	Review will be done in 3 phases: 1. Internal analysis and preparation of issue papers.	State Controller's Office Public Employees' Retirement System Department of Financ
Subchapter 1 General Civil	·			Obtaining of public comment.	State Compensation Insurance Fund Office of Employee
Service Rules Articles 7-15 19,22,25	June 81	July 81		3. Decisions made on actions to be taken re-	Relations
Articles 1-4 20 - 21	Aug. 81	Oct. 81		garding rules under review.	
Subchapter 2 Particular Em- ployment Prac- tices					
Articles 1-16 Subchapter 3 Race and Color Discrimination	Aug. 81	Oct. 81			
Article 1 Subchapter 1	Aug. 81	Oct. 81			
Articles 5-6 16-18.6,23-24, 26	Nov. 81	Jan. 82			
Division V Chapter 1					
Articles 1-9	Aug. 81	Oct. 81			

AGENCY: Board of Pharmacy

AGENCY COORDINATOR: Claudia Klingensmith

ADDRESS: 1021 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5014

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16		·			
Chapter 17 California State Board of Pharmacy	Feb. 82	March 82	Board Newsletter Newspapers Radio Television Trade Journals Consumer Newsletters Advisory Committees	After completion of review process, regulatory changes to be considered at Board meetings to be scheduled.	Department of Health Services
			,		
					·

AGENCY: Board of Registration for Professional Engineers

AGENCY COORDINATOR: Denice E. Brue

ADDRESS: 1006 4th Street, 6th Floor, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5544

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 5 Board of Registration for Professional Engineers	Feb. 82	Apr. 82	Trade Journals Newspapers Newsletters Mailing List	Following completion of review process, regulatory changes will be considered at hearings scheduled for May and June of 1982.	Department of Consumer Affairs
	·				

Page 121

AGENCY: Public Employees' Retirement System

AGENCY COORDINATOR: Charles I. Gibbs

ADDRESS: P.O. Box 1953, Sacramento, CA 95809

TELEPHONE NUMBER: (916) 445-6867

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 1 Chapter 2 Subchapter 1 Employees' Retirement System Regulations Subchapter 2 Social Security (OASDHI) Regulations Subchapter 3 State Employees' Medical and Hospital Care Act Regulations	May 81 May 81	Nov. 81 Nov. 81	Notices will be posted in Agency offices and have been published in the "Retirement Newsletter" of February 1981. Written comments will be solicited February 1981 through April 1, 1981. Oral comments from interested parties may be presented at Board meetings during April and May 1981. State and local agencies and school districts will be notified of review plan.	Board meetings scheduled: April 22, 1981 at 10:30AM in Los Angeles May 19, 1981 at 1:30PM in Sacramento Written comments to Agency on or before April 1, 1981.	Federal Old Age and Survivors' Insurance Program. Public Employees' Medical and Hospital Care Act.

AGENCY: Structural Pest Control Board

AGENCY COORDINATOR: Mary Lynn Ferreira

ADDRESS: 1430 Howe Avenue, Sacramento, CA 95825

TELEPHONE NUMBER: (916) 924-2291

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 19 Structural Pest Control Board	June 81	Aug. 81	Mailing List Real Estate Trade Journals Notices to Licensees 3 Advisory Committees a. Pesticide Advisory Committee b. Technical Advisory Committee c. Education Advisory Committee	Advisory Committees with support staff will review regulations and make recommendations to the Board.	Environmental Protectio Agency Department of Food and Agriculture Department of Health Department of Real Estate

AGENCY: Tax Preparer Program

AGENCY COORDINATOR: Donald Procida

ADDRESS: 1430 Howe Avenue, Sacramento, CA 95825

TELEPHONE NUMBER: (916) 920-6101

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 32 Division of Consumer Services - Tax Preparers	Oct. 81	Dec. 81	Newspapers Legal Journals Notices to Consumer Groups	After completion of review process, regulatory changes to be considered at Board meetings to be scheduled.	Department of Consumer Affairs
				·	
			· ·		
					•

AGENCY: State Teachers' Retirement System

AGENCY COORDINATOR: Nancy S. Dolder

ADDRESS: P.O. Box 15275-C, Sacramento, CA 95813

TELEPHONE NUMBER: (916) 920-7005

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 5					
Part III			Mailing Lists	Representatives of the following organizations	County Superintendents of Schools in all 52
Chapter 1				will be solicited to pro-	Counties
Article l Definitions	June 81	Sept. 81		vide comments: California Teachers'	
Article 2 Rules of Proce- dure	June 81	Sept. 81		Association American Federation of Teachers California Retired	
Article 3 General Provisions of Administration	June 81	Sept. 81		Teachers' Association Associated California School Administrators	
Article 4 Tax Sheltered Annuity Contri- butions	June 81	Sept. 81			
Article 7 Election of Options	June 81	Sept. 81			
Article 8 Rates	June 81	Sept. 81			
Article 10 Dependents	June 81	Sept. 81			
Article 11 Conflict of Interest Code	June 81	Sept. 81			

Page 125

AGENCY: Department of Veterans Affairs

AGENCY COORDINATOR: Dean Stiles

ADDRESS: 1227 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-6437

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 12					ı
Division 2			Mailing List	All existing regulations	Department of Real
Subchapter 1 Farm and Home Purchases	Apr. 81	June 81		are to be repealed. New proposed regulations will be prepared and public comment will be solicited.	Estate Department of Savings and Loan California Housing
Subchapter 2 Veterans' Educa- tional Institute	Apr. 81	June 81	·	Final regulations will be adopted pursuant to the requirements of the Administrative Procedure	Finance Agency
Subchapter 3 Veterans' Depen- dents Educational Assistance	Apr. 81	June 81		Act.	
Subchapter 4 Service and Coor- dination	Apr. 81	June 81			·
Subchapter 5 Veterans' Institu- tions	Apr. 81	June 81			
Subchapter 6 Conflict of Interest Code	Apr. 81	June 81			
		,	,		
	·				
			,		
	1			•	

AGENCY: Board of Examiners in Veterinary Medicine

AGENCY COORDINATOR: Gary Hill

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-4070

Page1<u>26</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 20 Board of Examiners in Veterinary Medicine Articles 1,2,3,5, 6,7	Nov. 81	Dec. 81	Mailing List Newsletters	Following completion of the review process, regulatory changes will be considered at three regularly scheduled Board hearings.	Horse Racing Board Department of Food and Agriculture Department of Education
Article 4	Feb. 82	Mar. 82			
		·			
	٠.				

AGENCY: Board of Vocational Nurse and Psychiatric Technician Examiners

AGENCY COORDINATOR: Marcia Peterson

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-0793

Page <u>127</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16 Chapter 25 Board of Vocational Nurse and Psychi- atric Technician Examiners of the State of California			Mailing List Newsletters	After completion of review process, regulatory changes will be considered at hearings to be scheduled.	Department of Mental Health Department of Develop- mental Services Department of Health Services
Subchapter 1 Vocational Nurses Subchapter 2 Psychiatric	Apr. 82	May 82 May 82		·	·
Technicians					
					•
	•				

AGENCY: Board of Corrections

AGENCY COORDINATOR: Edgar Smith

ADDRESS: 1924 T Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5073

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 15 Division I Subchapter 3 County Justice System Subvention Program Appeal Procedure Subchapter 4 Minimum Standards for Local Detention	Sept. 81	Oct. 81	sheriffs, chiefs of police, jail administrators, county administrative officers and inmate advocacy groups. Board of Corrections Quarterly Newsletter Notice Register Public Meeting scheduled for September 16, 1981.	Review plan time frame coincides with the regular mandated biennial review of minimum standards for Local Detention Facilities regulations. By mid September 1981 all jails in the state will have been inspected according to these existing regulations. Any problems that exist with these regulations will be better identified after the inspection is completed.	Services Office of State Fire Marshal Building Standards Commission

AGENCY: Department of Corrections

AGENCY COORDINATOR: Ben Williams

ADDRESS: 630 K Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5916

OAL Master Plan For Review of Existing Regulations

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
ADMINISTRATIVE CODE UNDER REVIEW Title 15 Division 3 Chapter 1 Subchapter 1 Inmate Rules Subchapter 2 Inmate Resources Articles 1-6 and 8-9 Article 7 Subchapter 3 Inmate Activities Subchapter 4 General Institution Regulations Subchapter 5 Personnel Subchapter 6 Parole	COMMENT ENDING DATE Jan. 81	OF		Administrative Bulletins and Notices distributed during December 1980 notifying inmates, employees, administrators and interested members of the public of the review. Meetings to be held at each of the 12 institutions to solicit input. All comments to be reviewed.	As required.
				·	

AGENCY: Narcotic Addict Evaluation Authority

AGENCY COORDINATOR: Marilyn Sellers

ADDRESS: P.O. Box 1841, Norco, CA 91760

TELEPHONE NUMBER: (714) 737-2683

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 15 Division 5 Chapter 1 General Chapter 2 Hearings Chapter 3 Outpatient Supervision Chapter 4 Appeals Chapter 5 Discharge of Narcotic Addicts	May 81 May 81 May 81 May 81	June 81 June 81 June 81 June 81 June 81	Newspapers Los Angeles Times Riverside Press Enterprise Sacramento Bee Mailing Lists	All comments received from the public, Department of Corrections, residents and outpatients committed to Civil Addict Program will be reviewed in addition to the Board's own internal review of existing regulations. (All written comments must be received by May 1, 1981.)	Corrections

AGENCY: Board of Prison Terms

AGENCY COORDINATOR: Joan W: Cavanagh

ADDRESS: 545 Downtown Plaza, Ste. 200, Sacto, CA 95814

TELEPHONE NUMBER: (916) 322-6729

Page <u>131</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 15					
Division 2			Mailing List	Public comment will be	Department of
Chapter 1 General	May 81	June 81	Task Force comprised of representatives from the Office of the Attorney	solicited, reviewed and considered by the Board at public hearings	Corrections
Chapter 2 Term Decisions	May 81	June 81	General, California Dist- rict Attorneys' Associa-	scheduled for May 1981 in Los Angeles, Sacramento	·
Chapter 3 Parole Release	May 81	June 81	tion, State Public Defend- er, Prisoners Union and the State BAR Commission	and two prison locations to be selected.	
Chapter 4 Postponement or Rescission of Release	May 81	June 81	on Corrections. Internal Administrative Directives		
Chapter 5 Parole Supervision	May 81	June 81			-
Chapter 6 Parole Revocation	May 81	June 81			
Chapter 7 Executive Clemency	May 81	June 81			·
					·

AGENCY: Department of the Youth Authority

AGENCY COORDINATOR: Darrell Bray

ADDRESS: 4343 Williamsbourgh Dr., Sacto, CA 95823

TELEPHONE NUMBER: (916) 322-1278

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 15			·	·	
Division 4			Mailing Lists	Advisory Committees repre-	Board of Corrections
Chapter 1 General Provisions	Mar. 81	May 81	Notice Register	senting Chief Probation Officers Association and Detention Facility Managers	Youthful Offender Parole Board
Chapter 1.5 Assistance to Courts	Mar. 81	Apr. 81		Association will review Chapter 2, Subchapters 3 and 4.	
Chapter 1.6 Commitment to Youth Authority	Mar. 81	Apr. 81		Advisory Committee representing 57 counties involved in County Justice System Subvention Program will re-	
Chapter 1.7 Return to Court	Mar. 81	Apr. 81		view Subchapter 5.	
Chapter 1.8 Extended Confine- ment	Mar. 81	Apr. 81		Public hearings for final regulatory changes are scheduled for late 1981 to early 1982.	
Chapter 2 Prevention and Community Corrections					
Subchapters 2,6	Mar. 81	May 81			
Subchapters 1,3, 4,5,7,8	Aug. 81	Sept. 81			
Chapter 3 Institutions and Camp Services	Mar. 81	June 81			
Chapter 4 Parole Services	Mar. 81	May 81			

AGENCY: Youthful Offender Parole Board

AGENCY COORDINATOR: Darrell Bray

ADDRESS: 4241 Williamsbourgh Drive, Sacto, CA 95823

TELEPHONE NUMBER: (916) 322-1278

Chapter 5 Subchapter 1 General Provisions Subchapter 2 Board Pules Notice Register Newspapers: Los Angeles Journal Sacramento Union Notice Register Newspapers: Los Angeles Journal Sacramento Union Notice Register Newspapers: Los Angeles Journal Sacramento Union Proview process, public hearings on final regulatory changes are proposed for the week of September 14, 1981 in Sacramento. De D	COORDINATION	REVIEW PROCESS	METHODS OF PUBLIC INVOLVEMENT	STATEMENT OF COMPLETION	PUBLIC COMMENT ENDING DATE	ADMINISTRATIVE CODE UNDER REVIEW
Chapter 5 Subchapter 1 General Provisions Subchapter 2 Board Rules Relating To Wards In Institutions And Camps Subchapter 3 Board Rules Relating to Parole Subchapter 4 Discharge of Mar. 81 June 81						Title 15
Subchapter 1 General Provisions Subchapter 2 Board Rules Relating To Wards In Institutions And Camps Subchapter 3 Board Rules Relating to Parole Subchapter 4 Discharge of Mar. 81 June 81 June 81 Newspapers: Los Angeles Journal Sacramento Union Newspapers: Los Angeles Journal Sacramento Union Ings on final regulatory changes are proposed for the week of September 14, 1981 in Sacramento. Del Congression of the week of September 14, 1981 in Sacramento. Del Congression of the week of September 14, 1981 in Sacramento.	Department of the					Division 4
Subchapter 1 General Provisions Subchapter 2 Board Rules Relating To Wards And Camps Subchapter 3 Board Rules Relating to Parole Subchapter 4 Discharge of Mar. 81 June 81 Los Angeles Journal Sacramento Union Changes are proposed for the week of September 14, 1981 in Sacramento. De Discharge Journal Sacramento Union Mar. 81 June 81 June 81 June 81 June 81 June 81 June 81	Youth Authority Department of					Chapter 5
Board Rules Relating To Wards In Institutions And Camps Subchapter 3 Board Rules Relating to Parole Subchapter 4 Discharge of	Corrections Department of Mental Health	changes are proposed for the week of September 14, 1981 in Sacramento.	Los Angeles Journal	June 81	Mar. 81	General Provi-
Board Rules Relating to Parole Subchapter 4 Mar. 81 June 81 Discharge of	Department of Developmental Service: Board of Corrections			June 81	Mar. 81	Board Rules Relating To Wards In Institutions
Discharge of				June 81	Mar. 81	Board Rules Relating to
				June 81	Mar. 81	Discharge of
		·				

AGENCY: Agricultural Labor Relations Board

AGENCY COORDINATOR: Jorge Leon

ADDRESS: 915 Capitol Mall, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-6913

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 8 Part II	July 81	Aug. 81	Mailing list.	The Board is in the process of determining the	As required.
Chapter 1 Definitions Chapter 2 Unfair Labor Prac- tice Regulations	July 61	Aug. 01		need for issue papers and the need for public discussion of access rules under Chapter 9.	
Sections 20200- 20266	July 81	Aug. 81			
Sections 20268- 20298	Aug. 81	Sept. 81			
Chapter 3 Procedure Under Chapter 5 - Repre- sentation of Em- ployees					
Sections 20300- 20363	Sept. 81	Oct. 81			
Sections 20365- 20393	Oct. 81	Nov. 81			
Chapter 4 Service and Filing of Papers	Nov. 81	Dec. 81			
Chapter 5 Certification and Signature of Documents	Nov. 81	Dec. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 6 Availability of Board Materials and Documents	Nov. 81	Dec. 81			
Chapter 7 Prohibited Communi- cations	Nov. 81	Dec. 81			
Chapter 8 Practice Before the Board	Nov. 81	Dec. 81			
Chapter 9 Solicitation by Non-Employee Organizers	Dec. 81	Jan. 82	·		
Chapter 10 Symbols or Emblems	Jan. 82	Feb. 82			
Chapter 11 Emergency Meetings	Jan. 82	Feb. 82			
				·	

AGENCY: Air Resources Board

AGENCY COORDINATOR: David Nawi

ADDRESS: 1102 Q Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-2884

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 13					
Chapter 3			Agency Mailing List Newspapers	Based on its internal re- view and public comments	As required.
Subchapter 1 Motor Vehicles Pollution Control Devices			Newspapers	received, the staff will propose to the Executive Officer action with respect to each regulation. Action	
Articles 1,2	May 81	Mar. 82		on each regulation will be determined by the Executive	·
Articles 3,4	June 81	Jan. 82		Officer.	
Article 5	June 81	Feb. 82			,
Article 6	Oct. 81	Apr. 82			
Subchapter 2 Vehicle Emission Standards and Surveillance Test- ing	July 81	Apr. 82			
Subchapter 3 Highway and Manda- tory Inspection Emission Standards	Aug. 81	Feb. 82			
Subchapter 4 Pollution Control Devices	Sept. 81	Mar. 82	•	,	
Subchapter 5 Motor Vehicle Fuels	Sept. 81	Apr. 82			
	•				·

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 6 Exhaust Emission Control Devices	Aug. 81	Feb. 82			
Subchapter 7 Fill Pipes	Sept. 81	Mar. 82		· .	•
Citle 17					
Chapter 1					·
Subchapter 1 Administrative	Mar. 81	July 81			
Subchapter 1.5 Air Basin and Air Quality Standards					·
Article 1	June 81	Sept. 81			
Article 2	Apr. 81	July 82			
Subchapter 2 Agricultural Burn- ing	Apr. 81	Oct. 81			
Subchapter 3 Subventions	Aug. 81	Jan. 82			
Subchapter 4 Air Pollution Records	May 81	Sept. 81			
Subchapter 5 Emission Data	June 81	Jan. 82			
Subchapter 6 Abrasive Blasting	Oct. 81	Feb. 82			
Subchapter 7 Emergency Meetings	Mar. 81	July 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COURDINATION
Subchapter 8 Nonvehicular	July 81	Jan. 82			
Emission Standards Subchapter 9 Conflict of Interest Code	Aug. 81	Jan. 82			·.
					· .
	·				
τ.				•	36

Page 139

AGENCY: California Arts Council

AGENCY COORDINATOR: Todd Kaufman

ADDRESS: 2022 J Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-8917

ADMINISTRATIVE CODE UNDER REVIEW		TEMENT OF METHODS OF PLETION PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 3 Chapter 2 Grant Application Criteria and Procedure of the California Arts Council	Nov. 81 Jan.	Public meetings: San Diego 7-10-81 Eureka 8-14-81 Sacramento 9-18-81 Sacramento 9-19-81 Notices will be published in the following journals and the newsletter State of the Arts: Art Matters CTC News	Issues of high public interest will be discussed in issue papers and digests in June, July, and August of 1981 issues of State of the Arts. Oral testimony at meetings should be submitted in writing. All written comments due in Agency on or before November 1, 1981.	itation

Page 140

AGENCY: Board of Chiropractic Examiners

AGENCY COORDINATOR: Susan Rickard

ADDRESS: 921 11th Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-3244

Dec. 81	Feb. 82	Chiropractic Journals Mailing List Notice Register Two Advisory Committees	Summary of existing regulations will be submitted and comments solicited from industry and public, and evaluation reports will be submitted to Board from Advisory Committees and will be available to public.	Board of Medical Quality Assurance
Dec. 81	Feb. 82	Mailing List Notice Register	tions will be submitted and comments solicited from industry and public, and evaluation reports will be submitted to Board from Advisory Committees and will be	Quality Assurance
			•	
			·	

AGENCY: California Community Colleges

AGENCY COORDINATOR: Tom Nussbaum

ADDRESS: 1238 S Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-8752

Page <u>141</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC STATES COMMENT OF ENDING DATE COMPLE	METHODS OF	REVIEW PROCESS	COORDINATION
	June 81 June 81 June 81 June 81 July June 81 Jucc. June 81 July	Mailing list Mailing list	Regulation changes will be considered by the Board of Governors at full Board hearings San Diego November 5-6 Los Angeles November 20 (Board Committees) San Jose December 10-11 (Full Board) Meetings for 1982 have not been determined.	COORDINATION Community College Districts Department of Finance Postsecondary Education Commission

Page 142

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 7	June 81	July 82			
Chapter 8	June 81	Dec. 81		·	·
Division 7 Special Programs					
Chapter 1	June 81	Dec. 81			
Chapters 2,3,4,5	June 81	July 82			
Division 8 Construction	,				·
Chapter 1	June 81	Dec. 81	·		
Chapter 2	June 81	July 82			•
Division 9 Continuing Education	June 81	July 82			
Division 10 Community College Administration				·	
Chapter 2	June 81	Nov. 81	·		
Chapter 2.5	June 81	July 82			
Chapter 3	June 81	Aug. 81		_	
Chapter 4	June 81	July 82			
*					•

AGENCY: Board of Control

AGENCY COORDINATOR: Fred Buenrostro

ADDRESS: 926 J Street, Ste. 300, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 323-3564

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 2 Chapter 1 Article 1 General Provision Article 2 Appropriations Article 3 Presentation of Claims to the Controller Article 6 Payroll Claims Article 6.1 State Contributions Under State Employees' Medical and Hospital Care Act Article 6.6 Property Tax Relief Article 7 Invoice Claims	July 81 July 81 July 81 July 81	Nov. 81 Nov. 81 Nov. 81 Nov. 81 Nov. 81	A two-month public comment period is planned mailing lists to be used to solicit comments Public Meetings will be scheduled as needed. Entities to be included in review: Local Victim/Witness Centers Private Citizens and Businesses Legal Organizations Employee Organizations	The Department of General Services will in- clude Title 2, Division 2, Chapter 1, Article 17 in a public meeting on June 18, 1981. Issue papers will be developed prior to solici- ting public comments.	Universities and Colleges, Chancellor's Office Department of Finance Governor's Office Computer Assisted Personnel Transport Index System (CAPTIS) Department of General Services State Personnel Board Public Employees' Retirement System State Controller's Offi Merit Award Board

Page 144

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 4 Presentation of Claims to the State Board of Control	July 81	Oct. 81			
Article 5 Claims of Persons Erroneously Con- victed of Felonies	July 81	Oct. 81			
Article 6, §663 Payroll Deductions for Charitable Contributions	July 81	Oct. 81			
Article 5.5 Indemnification of Private Citizens	June 81	Oct. 81			
Article 5.6 Indemnification of Victims of Crimes Occurring Prior to 7-1-74	June 81	Oct. 81			
Article 5.7 Indemnification of Victims of Crimes Occurring on or After 7-1-74	June 81	Oct. 81		·	
Article 8 Traveling Expenses	Aug. 81	Jan. 82			
Article 8.1 Moving and Reloca- tion Expenses	Aug. 81	Jan. 82			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 8.5 Community Rela- tions Expenses- State University and Colleges Presidents	Aug. 81	Jan. 82			
Article 8.6 Community Relations Expenses of the Chancellor of the California State University and Colleges	Aug. 81	Jan. 82			
Article 10 Agents Transport- ing Prisoners, Patients and Wards	Aug. 81	Jan. 82			
Article 15 Use of State-Owned Vehicles	Aug. 81	Jan. 82			
Article 16 Valuation of Employee Housing	Aug. 81	Jan. 82			
Article 16.1 Meals at State Agencies	Aug. 81	Jan. 82			
Article 17 Purchase of Sup- plies or Equip- ment for State Agencies	Aug. 81	Jan. 82			

Page <u>146</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 19 Overpayments	Aug. 81	Jan. 82			
Article 20 Miscellaneous Reimbursement	Aug. 81	Jan. 82			
Article 14 Employee Merit Award Program	May 81	Sept. 81			
	÷		_		
		4.			
					·

AGENCY: California Educational Facilities Authority

AGENCY COORDINATOR: Elizabeth Whitney

ADDRESS: 915 Capitol Mall, Room 110, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-9603 (213) 620-4161

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 12			Notices published in San Francisco Examiner,	Regulatory changes will be determined through	Department of Finance State Banking Department
Article 1 General	May 81	June 81	Los Angeles Times, Sacramento Union	contact with the Bond Counsel, Investment Bank-	State Treasurer
Article 2 Definitions	May 81	June 81		ers and College Adminis- trators.	
Article 3 Procedures Relating to the Authority of Officers and Members	May 81	June 81			
Article 4 General Provisions Relating to Auth- ority Actions	May 81	June 81			
·					
	,				
			,		

AGENCY: California Energy Commission

AGENCY COORDINATOR: Arlene Ichien

ADDRESS: 1111 Howe Avenue, Sacramento, CA 95825

TELEPHONE NUMBER: (916) 920-6257

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 20					
Chapter 2			Mailing Lists	Issue papers will be pre-	Coordination with
Subchapter 7 Administration	Feb. 81	June 81	Major Newspapers	pared and issued at two- week intervals for each subchapter.	state agencies will be accomplished by distributing issue
Subchapter 6 Environmental Protection	Mar. 81	June 81		45-day response time will be allowed for issue papers.	papers for review and comment.
Subchapter 4 Load Management				Recommendation of the Pub- lic Advisor will govern the	
Article 5	Mar. 81	June 81		scheduling of public meet-	
Subchapter 3 Data Collection	Mar. 81	June 81		ings.	·
Subchapter 8 Solar Energy	Apr. 81	June 81		·	
Subchapters 1,2,5 Cite Certification	Apr. 81	June 81			
Subchapter 7 Public Advisor					
Article 3	Apr. 81	June 81			
Subchapter 4 Insulating Mater- ials Standards			•		
Article 3	May 81	June 81			
Article 4 Appliance Effi- ciency Standards	May 81	June 81			·

AGENCY: Board of Equalization

AGENCY COORDINATOR: Janice Masterton

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-6479

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 18					
Chapter 1 Property Taxes Chapter 2 Business Taxes	Dec. 81	July 82	Newspaper State Board of Equalization Tax Information Bulletin California Register Mailing List to: govern- mental agencies, Califor- nia taxpayers by selected industries categories, interested parties Task Force will be formed from taxpayer groups and the California Assessors' Association Public Hearings will be	After the completion of the review process, hearings for the adoption of changes to regulations will be scheduled.	State Controller's Office Department of Motor Vehicles
			scheduled from adopted Board meeting dates		
				,	

AGENCY: Fair Political Practices Commission

AGENCY COORDINATOR: Jeanne Pritchard

ADDRESS: 1100 K Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-5901

					i
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2					
Division 6 Chapter 1 General	May 81	May 81	The <u>FPPC Bulletin</u> and the <u>Lobbyist Bulletin</u> will carry notices regarding review process.	FPPC monthly meetings will include review of existing regulations.	Secretary of State Attorney General Franchise Tax Board State Legislators
Chapter 3 Fair Political Practices Commis- sion	May 81	May 81	The public is invited to request to be placed on the mailing list. Comments will also be	Issue papers available to public on request prior to or at public meetings.	Common Cause Consumer's Union All City and County Agencies
Chapter 9 Incumbency	May 81	May 81	solicited from specific groups (see Coordination).		League of California Cities County Supervisors'
Chapter 10 Auditing	May 81	May 81			Association of California
Chapter 2 Section 18227 Filing Officer	May 81	May 81			Institute of Govern- mental Advocates County and City Clerks Associations
Section 18241 Culmination Period for Mass Mailings	May 81	May 81			Political Campaign Consultants Law Firms
Chapter 6 Lobbyists	June 81	June 81			
Chapter 2 Section 18202 Quasi-Legislative Administrative Action	June 81	June 81			

Section 18239 "Lobbyist"; Definition of Terms Used in Government	June 81	June 81		
Code Section 82039				
Chapter 7 Conflicts of Interest; General Prohibition	Aug. 81	Sept. 81		• •
Chapter 2				
Section 18233 Leasehold Interest	Aug. 81	Sept. 81		
Section 18234 Reporting Inter- ests in Trusts	Aug. 81	Sept. 81		e e e e
Section 18235 Reporting Inter- ests in a Blind Trust	Aug. 81	Sept. 81		·
Section 18249 State Agency	Aug. 81	Sept. 81		
Chapter 4 Campaign Disclosure	Nov. 81	Dec. 81		
Chapter 2				
Section 18215 Contribution	Nov. 81	Dec. 81		
Section 18225 Expenditure	Nov. 81	Dec. 81		•
				,

AGENCY: Department of Food and Agriculture

AGENCY COORDINATOR: Mr. Francis Stoffels

ADDRESS: 1220 N Street, Rm. 153-B, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-4888

Page <u>152</u>

Title 3 Chapter 1 Administration				I control of the cont	
Subchapters 1,2 General Examination for County Officers and Employees Subchapter 3 California Environmental Quality Act Chapter 2 Animal Industry Subchapter 3 Livestock Identification Subchapter 4 Meat Inspection Subchapter 5 Poultry Inspection Subchapter 2 Livestock Disease Control Subchapter 6 Drugging of Horses	Nov. 81 May 81 May 81 June 81	Dec. 81 Dec. 81 June 81 June 81 June 81 July 81 July 81	Subchapter 1,2 (Chpt. 1) Mailings to: Advisory Committee of County Commissioners and County Sealers Subchapter 3 (Chpt. 1) Newspapers in Los Angeles, Fresno, San Francisco, and Sacramento Chapter 2 Industry Publications Subchapter 1 (Chpt. 3)(Gp.1) Canning Tomato Advisory Committee Subchapter 1 (Chpt. 3)(Gp.3) Mailings to User Groups, requesting publishing space in Pacific Egg and Poultry Association Newsletter Subchapter 1(Chpt.3)(Gps.5, 10, Art. 1) Mailings to interested parties Press Releases	Subchapter 1(Chpt.3)(Gps.3, 9) Meetings scheduled in	Subchapters 1,2(Chpt, 1) County Departments of Agriculture County Departments of Weights and Measures State Food and Agriculture Public Employment and Affirmative Action Division State Personnel Board Subchapter 3(Chpt. 1) All State Agencies Subchapters 1,2,3,4,5,6,7,8 (Chpt. 2) Department of Food and Agriculture - Exclusion and Detection- Pesticide Enforcement U.S. Department of Agriculture-Bureau of Animal Health- Bureau of Meat Inspection Department of Health Services Bureau of Milk and Dairy Foods Control

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 7 Control of Diseased Animals Subchapter 1 Dairies Subchapter 8 Animal Biologics Chapter 3 Economics Subchapter 1 Group 9 Poultry and Rabbit	June 81 Oct. 81 Oct. 81	July 81 Nov. 81 Nov. 81	Subchapter 1(Chpt.3)(Gp.10, Art. 2) Public Meeting on Garlic and Onion Dehydration held in Sacramento on 3/10/81. Subchapter 3(Chpt. 2) American Dairyman Sacramento Bee Subchapter 1 (Chpt. 4) (Gp. 1) General Circulation Newspapers California Fertilizer	Subchapter 1(Chpt. 3)(Gp.1) The Canning Tomato Advisory Committee will solicit public comments at a meeting to be held in Sacramento. Issue paper is available upon request Subchapter 1(Chpt. 3)(Gp.3) Review of regulations will take into consideration U.S. Department of Agriculture's plan to change federal egg and poultry regulations.	Committees and Organ- izations County Public Health Agencies Numerous Private Busi- nesses and Public Entities
Group 1 Canning Tomatoes	May 81	July 81	Association Monthly Bulletin Subchapter 2 (Chpt. 4)	Meetings scheduled for August: One in Los Angeles	to be held in Sacra- mento. Issue paper is available upon request.
Group 3 Eggs Group 5 Grapes for By- products	Aug. 81 Nov. 81	Sept. 81 Dec. 81	(Gp. 2) Public Notice will be published in local news- papers Subchapter 2 (Chpt. 4)	One in Sacramento Subchapter 1(Chpt.3)(Gps.5, 10) Public meetings will be scheduled in November 1981 in Sacramento.	Subchapter 1(Chpt. 3) (Gps. 3,9) Shell Egg Advisory Committee County Agricultural
Group 10, Art. 1 Wine Grape In- spection Group 10, Art. 2 Garlic and Onions for Dehydration	Nov. 81	Dec. 81	(Gp. 3) Notices in California Arizona Farm Press, California Farmer, Sacramento Union, Fresno Bee, California Grain	Subchapter 3(Chpt.3) Issue papers available upon request. Subchapter 4(Chpt.4)(Gp.1) Mailings will include	Commissioners U.S. Department of Agriculture Subchapter 1(Chpt.3) (Gps.5, 10, Art. 1) Grape Inspection
Subchapter 3 Stabilization and Marketing of Milk and Dairy Products	May 81	June 81	and Feed Conveyor Subchapters 3,4,5(Chpt. 4) Noticing in: General Newspapers concerned with particular regulations; also watch California	copies of notice, regula- tions' objectives, and statements of purpose. Public meetings will be scheduled throughout California and publicized	Advisory Committee Growers' Grape Inspec- tion Advisory Committee County Agricultural Commissioners

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Chapter 4 Plant Industry Subchapter 1 Chemistry Group 1 Fertilizing Materials Subchapter 2 Field Crops Group 2 Commercial Feed Group 3 Grain and Commodity Inspection Subchapter 3 Entomology and Plant Quarantine Group 1 Apiaries Group 2 Nursery Stock Group 4 Plant Quarantine Group 5 Insect Pest Control	June 81 Sept. 81 May 81 June 81 June 81 July 81 July 81	Aug. 81 Oct. 81 June 81 July 81 Aug. 81 Aug. 81	Farmer for notices. Mailings will include notice, regulations be- ing reviewed, and state- ment of purpose and objections. Subchapter 3(Chpt. 4) Noticing through press releases and publication in general circulation newspapers statewide. Reviews will be included on Pesticide Advisory Committee agendas. There will be a statement of purpose and objectives on each regulation to be reviewed. Chapter 8 (Title 4) Agency is soliciting written comments. Notices will be published in general circulation newspapers and numerous trade journals includ- ing: Weighing and Measurement Magazine, National Scaleman's Association Weighlog, Oil Express, National Petroleum News, California Grocers' Journal, Pacific Coast	Chapter 8 (Title 4) Issue papers will go with mailings to affected groups, agencies and industries. Chapter 3 (Title 3) Will publish in production area newspapers and trade magazines such as California Farmer and Packer.	Subchapter 3(Chpt.3) Issue papers available upon request. Subchapter 1(Chpt.4) (Gp. 1) Fertilizer Advisory Board California Fertilizer Association University Staff California Farm Advisors California Farm Bureau Federation Association of California Nurserymen Subchapter 2(Chpt.4) (Gp. 2) Food and Agriculture Measurement Standards Animal Health Commercial Fertilizer University of California - Davis Feed Advisory Boards Subchapter 2(Chpt. 4) (Gp. 3) U.S. Department of Agriculture - Federal Grain Inspection Ser. California Grain and Feed Association California Farm Bureau Association

			•	•	
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 5 Rodent and Weed Control and Seed Inspection Group 1 One-Variety Cotton District Group 3 Seed Inspection Group 4 Weed-Free Areas and Weed Eradication Areas Group 6 Noxious Weed Species Subchapter 4 Plant Pathology Group 1 Host-Free Districts and Periods Group 2 Meyer Lemon Chapter 4 Plant Industry Subchapter 3 Entomology and Plant Quarantine Group 3 Pest Control Operations	June 81 June 81 Oct. 81 Oct. 81	July 81 July 81 Nov. 81 Nov. 81	Meat Association, California Van and Storage Association Newsletter, and California Canners' League Newsletter Chapter 3 (Title 3) Agency is soliciting written comments. A meeting planned for each commodity. Agencies will be notified of meetings and pertinent informa- tion, issue papers. Article 7 is now under review for extensive changes. Hearings are underway March 81 with findings pending.		California Association of Wheat Growers Northern California Grain Exchange Los Angeles Grain Exchange Private Businesses: Continental Grain Cargill, Inc. Koppel Bulk Terminals Garnac Grain Pacific Vegetable Oil Corporation Subchapters 3,4,5 (Chpt.4 Department of Food and Agriculture - Agricultural Chemicals Division California Agricultural Commissioners' Assoc. California Farm Advisors California Farm Advisors California Association of Nurserymen Agricultural Officials From Other States U.S. Department of Agriculture Citrus Producers Numerous Pest Control Boards - List Available through Dept. Subchapter 3(Chpt.4) U.S. Environmental Protection Agency County Agricultural Commissioners

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 1 Definitions	May 81	June 81			State Board of Food and Agriculture
Article 2 Agricultural Pest Control Operator License	May 81	June 81			Committees: State Environmental Hazards Assessment Committee Health and Safety Advisory Group
Article 3 Permits to Operate With- out a License	May 81	June 81		·	Regional Accredita- tion Committees Agricultural Pest Control Advisory
Article 4 Aircraft Pilots	May 81	June 81			Committee Pesticide Registration and Evaluation
Article 8 Agricultural Pest Control Advisers	May 81	June 81			Committee Pesticide Advisory Committee Various Federal Agen-
Article 9 Pesticide Dealers	May 81	June 81			cies including Departments of the Interior, Army, Food and Drug
Article 5 Operations for Hire	Aug. 81	Sept. 81			Government of Mexico Chapter 8 (Title 4)
Article 6 Pest Control Operations	Aug. 81	Sept. 81			Environmental Protec- tion Agency National Bureau of Standards
Article 7 Recommenda- tions and Usage	Aug. 81	Sept. 81	,		Consumer Affairs Air Resources Board County Sealers of Weights and Measures
Article 10 Storage, Trans portation and Disposal	Nov. 81	Dec. 81			Chambers of Commerce

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 1 Chemistry Group 2 Economic Poisons Articles 1-11	July 81	Aug. 81			Chapter 3 (Title 3) Department of Food and Agriculture Measurement Standards Shipping Point Inspection
Articles 20-23 Title 4 Chapter 8 Division of Measure- ment Standards	Nov. 81	Dec. 81			Direct Marketing Program Department of Health Services, Food and Drug Section County Agricultural Commissioners
Subchapter 4 Tolerances and Specifications for Commercial Weighing and Measuring Devices	July 81	Aug. 81			
Subchapter 6.1 Device Repairmen	July 81	Aug. 81			
Subchapter 8 Submeter Billing for Utility Services	July 81	Aug. 81			
Subchapter 1 Automotive Pro- ducts Specifica- tions	Nov. 81	Dec. 81			
Subchapter 2 Standards of Weights and Measures	Nov. 81	Dec. 81			

vetandautemmataum dissolvitomis (initialis vetantautem vetantautem temperaturus vetantautem vetantaute	1	T			Γ
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 2.1 Package Inspection Under Federal Standards of Accuracy	Nov. 81	Dec. 81			
Subchapter 3 Polyethylene Commodities - Labeling and Testing	Nov. 81	Dec. 81			
Subchapter 5 Petroleum	Nov. 81	Dec. 81			
Subchapter 6 Administration	Nov. 81	Dec. 81			
Subchapter 7 Retail Sale of Motor Vehicle Fuel	Nov. 81	Dec. 81			
Title 3 Food and Agriculture Inspection Services, Fruit and Vegetable Quality Control					
Chapter 3 Economics					
Subchapter 1 Fruit and Vege- table Standardi- zation					
Group 6 Honey	Mar. 82	Apr. 82		> ₹	

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Group 4 Fresh Fruits, Nuts, and Vegetables					
Article 2 Inspection Procedures	June 15,82	June 30,82			
Article 4 Containers	June 15,82	June 30,82		•	
Article 32 Onions	June 15,82	June 30,82	*		
Article 42 Sweet Potatoes	Aug. 82	Sept. 82			
Article 5 Emergency Standards	Apr. 82	May 82			
Article 6.5 Direct Marketing	Apr. 82	May 82			
Article 17 Cantaloupes	June 15,82	June 30,82			
Article 28 Melons	June 15,82	June 30,82			
Article 6 Exempt Fruits Nuts, and Vegetables	Mar. 82	Apr. 82	,.		
Article 20 Celery	Apr. 82	May 82			·
	·				

Pagel60

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 24 Dates	Apr. 82	May 82			
Article 29 Mushrooms	Apr. 82	May 82			
Article 22 Citrus	Sept. 81	Feb. 82			
Article 11 Avocados	Apr. 82	May 82			•
Article 18 Carrots	Apr. 82	May 82	8-		
Article 23 .Corn	Apr. 82	May 82	,		
Article 40 Potatoes	Feb. 82	Mar. 82			
Article 13 Berries	Feb. 15,82	Feb. 28,82			·
Article 19 Cauliflower	Feb. 82	Mar. 82			
Article 27 Lettuce	Mar. 15,82	Mar. 31,82			
Article 31 Olives	Feb. 15,82	Feb. 28,82			
Article 30 Nectarines	Nov. 81	Dec. 81			
Article 34 Peaches	Nov. 81	Dec. 81	,		
Article 38 Plums, Prunes	Nov. 81	Dec. 81			
Article 35 Pears	Nov. 81	Dec. 81			
*	øt ·			\$	*

FRATIVE CODE R REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 16	Sep. 15,81	Sep. 30,81			
Cabbage	'				
Article 25	Sept. 81	Oct. 81			
Grapes,					
Raisins					
Article 36	Sep. 15,81	Sep. 30,81			
Peas				·	
Article 44	Sep. 15,81	Sep. 30,81			
Walnuts					
Article 33	June 81	July 81	1		
Green Onions,			1		
Shallots					
Article 10	Aug. 81	Sept. 81			•
Asparagus					
Article 41	June 81	July 81			
Quinces					
Article 14	July 81	Aug. 81			
Broccoli					
Article 39	Apr. 81	May 81			
Pomegranates					
Article 12	June 81	July 81		·	*
Bell Peppers					
Article 21	July 81	Aug. 81			
Cherries					
Article 1	Aug. 82	Sept. 82		Į.	
Markings					
Article 3	Aug. 82	Sept. 82		↓	
Definitions		04 00			
Article 37	Aug. 82	Sept. 82		•	
Persimmons	۸ 22	Sept. 82			
Article 43	Aug. 82	3ept. 62	1		
Tomatoes Article 8	May 81	June 81	1		
	riay or	Julie of			
Apricots	May 81	June 81	1		
Article 9	May of	Julie 01	1	1	
Artichokes					

Page 162

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 15 Brussel Sprouts	May 81	June 81		·	
Article 7 Apples	See Public	Involve-			
•					
				·	

AGENCY: California Health Facilities Commission

AGENCY COORDINATOR: Joseph Hafkenschiel,

ADDRESS: 717 K Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-2810

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 10			Mailing list to all hos-	Issue papers will be	Department of Health
Article l General Provisions	Sept. 81	Dec. 81	pitals, long-term care facilities, health plan- ners and health facility	developed and circulated for comments. The Attorney General's Office will re-	Services Local Health Systems
Article 2 Definitions	Sept. 81	Dec. 81	regulators. Advisory Council of the	view both the staff pre- pared material and the	Agencies Office of Statewide Health Planning and
Article 3 Procedures Relating to the Authority of Officers and Members	Sept. 81	Dec. 81	California Health Facil- ities Commission composed of hospital and long-term care administrators, labor union representa- tives, accounting experts	Advisory Council's final recommendations.	Development
Article 4 General Provisions Relating to Auth- ority Actions	Sept. 81	Dec. 81	and public members appointed by the Commission		
Article 5 Provisions Relat- ing to Small Busi- ness Financings	Sept. 81	Dec. 81			
Sections 7000- 7135					
				,	
				·	

AGENCY: California Housing Bond Committee

AGENCY COORDINATOR: Vern Afhby

ADDRESS: 915 Capitol Mall, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-3991 or 445-6562; (213) 642-6685

ADMINISTRATIVE CODE UNDER REVIEW	COMMENT	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 25 Chapter 6	June 81	Oct. 81		Review process under development.	
•				•	

Page 165

AGENCY: _State Lands Commission

AGENCY COORDINATOR: Gary Horn

ADDRESS: 1807 13th St., Sacramento, CA 95814

TELEPHONE NUMBER:	(916) 322-7807
-------------------	----------------

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 3			Mailing Lists	The Commission will survey	Local, federal, city
Chapter 1 State Property Operations			Public Meeting Newspapers	large landowners, both public and private, in search of common, prudent land management practices.	and county agencies will be contacted for their input into the review process.
Articles 1,2,2.5, 5,6,9	Mar. 81	May 81		A task force comprised of State Land Commission staff	
Articles 3,3.2, 3.3,3.4,3.5,4, 4.1	June 81	July 81		will also be used to review the regulations.	
Articles 6.5,10, 11	July 81	Sept. 81			
Articles 1.5,7, 8,12	Aug. 81	Nov. 81			
		9			
				·	

Page 166

AGENCY: California State Library

AGENCY COORDINATOR: Joan Frye

ADDRESS: P.O. Box 2037, Sacramento, CA 95809

TELEPHONE NUMBER: __(916) 322-0363

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
	Sept. 81 March 81 July 81 Aug. 81 Sept. 81 Sept. 81			Internal analyses, surveys, issue papers, impact and cost-benefit analyses will be conducted in addition to public meetings to determine regulation changes.	California Community Colleges Department of Correction

AGENCY: California Maritime Academy

AGENCY COORDINATOR: W. C. Black

ADDRESS: P.O. Box 1392, Vallejo, CA 94590

TELEPHONE NUMBER: (707) 644-5601

Page <u>167</u>

	ENDING DATE	OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 5			Nagara	Davidskam skanas váll ko	California State
Part VII Chapter 1		·	Newspaper California Maritime Daily Bulletin	Regulatory changes will be considered by the Califor- nia Maritime Academy Board	University and College
Subchapter 1 Employees	Apr. 81	June 81	Advisory Committee made up of Administrators,	of Governors meeting on May 13, 1981.	
Subchapter 2 Administration	Apr. 81	June 81	Faculty and Students Public meeting scheduled		
Subchapter 3 Student Fees	Apr. 81	June 81	for April 15, 1981 at 4:00 p.m. at the Califor- nia Maritime Academy Auditorium		
			v.		
	·	·			

AGENCY: Board of Osteopathic Examiners

AGENCY COORDINATOR: Gareth Williams

ADDRESS: 921 11th St., Suite 1201, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-4306

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 16					
Chapter 16 Board of Osteopathic Examiners	Aug. 81	Oct. 81	Mailing List Newspapers Trade Journals	After the conclusion of the review process, regu- latory changes will be considered at a public meeting to be scheduled.	Board of Medical Quality Assurance
				·	

AGENCY: Commission on Peace Officer Standards and Training

AGENCY COORDINATOR: Beverly Clemons

ADDRESS: 7100 Bowling Green Drive, Sacto, CA 95823

TELEPHONE NUMBER: (916) 445-4515

PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
		POST Bulletin	Initial review	California Highway
Apr. 81	May 81	letter POSTScripts POST Advisory Committee made up of law enforce- ment trainers, educators, chief and sheriff associ- ation representatives and	conducted at public meet- ing on March 12 in Sacra- mento. Findings and recommendations from this meeting are to be presented at a regularly scheduled meeting of the Commission on April 16 at the Sacra-	Patrol
		members of the general public. Public meeting held March 12, 1981.	mento Inn, Sacramento.	
		,		•
	COMMENT ENDING DATE	COMMENT OF ENDING DATE COMPLETION	COMMENT ENDING DATE OF COMPLETION METHODS OF PUBLIC INVOLVEMENT POST Bulletin Quarterly Commission Newsletter POSTScripts POST Advisory Committee made up of law enforcement trainers, educators, chief and sheriff association representatives and members of the general public. Public meeting held	Apr. 81 May 81 POST Bulletin Quarterly Commission News- letter POSTScripts POST Advisory Committee made up of law enforce- ment trainers, educators, chief and sheriff associ- ation representatives and members of the general public. Public meeting held POST Bulletin Quarterly Commission News- letter POSTScripts POST Advisory Committee made up of law enforce- ment trainers, educators, chief and sheriff associ- ation representatives and members of the general public. Public meeting held REVIEW PROCESS Initial review conducted at public meet- ing on March 12 in Sacra- mento. Findings and recommendations from this meeting are to be presented at a regularly scheduled meeting of the Commission on April 16 at the Sacra- mento Inn, Sacramento.

Board of Pilot Commissioners for the Bays of

AGENCY: San Francisco, San Pablo and Suisun

AGENCY COORDINATOR: Virginia Danielson

ADDRESS: World Trade Center, Room 342, San Francisco, CA 94111

TELEPHONE NUMBER: (415) 397-2253

Page <u>170</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 7				·	
Chapter 2 Article 1 Meeting of the Board	Mar. 81	June 81	All San Francisco Bar Pilots will be personally notified of the review	The prime tool of the review will be input from interested persons as well as the Bar Pilots themselves.	As required.
Article 2 Licensing of Pilots	Mar. 81	June 81	•	The comments received will be evaluated at the Board's March or April meetings.	
Article 3 Government of Pilots	Mar. 81	June 81		*	
		·			
	·				
				·	

Page 171

AGENCY: Public Employment Relations Board

AGENCY COORDINATOR: Steve Barber

ADDRESS: 923 12th Street, Suite 201, Sacto, CA 95814

TELEPHONE NUMBER: (916) 322-2531

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 8					
Part III Division 1 Public Employment Relations Board	Oct. 81	Apr. 82	Mailing List Newspapers Legal Reporter Services Trade Journals Agency House Organs	A preliminary public meeting will be held for each major group of regulations to provide basis for internal review. Following internal	All State Agencies
Division 2 Educational Employment Relations Act	Nov. 81	May 82	ngency nouse organs	review, a draft of regulatory changes will be available for public comment.	·
Division 3 State Employer- Employee Rela- tions Act	Dec. 81	June 82			
Division 4 Higher Education Employer-Employee Relations Act	Jan. 82	July 82			
			, ,		
,					

AGENCY: Seismic Safety Commission

AGENCY COORDINATOR: Helen Van Houte

ADDRESS: 1900 K Street, Suite 100, Sacto, CA 95814

TELEPHONE NUMBER: (916) 322-4917

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 19 Chapter 3	Dec. 81	Jan. 82	Questions on the Seismic Safety regulations may be addressed to the Commission.	The Commission welcomes questions on its regulations. It intends to submit to the Office of Administrative Law a Statement of Completion on or before January 31, 1982.	As required.
					•
			·		

AGENCY: California Student Aid Commission

AGENCY COORDINATOR: Barbara Schell

ADDRESS: 1410 Fifth Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-0880

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 5					O lis unio Protessondon
Part IV			Mailing List Notice Register	Program Managers will re- view existing regulations	California Postsecondary Education Commission
Chapter 1			Advisory Committees	and suggest changes.	Department of Education
Article l Definitions	June 81	Sept. 81		Director's Office will review suggestions from both staff and public input.	
Article 2 Applicant Qualifi- cations	June 81	Sept. 81		Final changes will be submitted to the Commission for approval.	
Article 3 Use of Cal Grants	June 81	Sept. 81			
Article 5 State Student In- centive Grant Program	June 81	Sept. 81			
Article 6 Conflict of Inter- est Code	June 81	Sept. 81			
Article 7 Statement of Purpose and Criteria for Approval as Processor for the Student Aid Application for California for Student Aid Commission Programs	June 81	Sept. 81			

Page <u>174</u>

		, and the second			
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Article 8 Approval or Dis- approval for Processing	June 81	Sept. 81			·
Article 9 The California Guaranteed Student Loan Program	March 81	April 81			
	·				
•	¢-			•	

AGENCY: Commission for Teacher Preparation and Licensing

AGENCY COORDINATOR: Walt Taylor

ADDRESS: 1020 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-0243

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Part VIII Chapter 1 Credentials Issued Under the Teacher Preparation and Licensing Law of 1970 Chapter 2 Rules of Procedure for the Commission Chapter 3 Committee of Credentials Chapter 4 Procedure for Application for Adding Authorization to, and Renewal of Credentials	Aug. 81	June 81 Sept. 81 Nov. 81	Mailing Lists Agency Newsletter Advisory Groups Public Meetings coinciding with scheduled Commission meetings.	Final regulatory changes will be considered by the State Board of Education at meetings to be held on: January 7-8 February 10-11 March 11-12 April 8-9 May 13-14 June 10-11 July 8-9 September 2-3 October 6-7 November 12-13 December 9-10 in Sacramento.	Department of Education (pursuant to Education Code 44232, every proposed adoption, amendment or repeal of regulations by Teacher Preparation and Licensing is subject to review and approval by the State Department of Education).
Articles 1,2,3	Nov. 81	Dec. 81			
Articles 4,5,6,7	Dec. 81	Jan. 82	,		
Chapter 5 Validation of Ser- vice Without a Credential	Dec. 81	Jan. 82			

AGENCY: California Transportation Commission

AGENCY COORDINATOR: Richard Donaldson

ADDRESS: 1120 N Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-1690

OAL Master Plan For Review of Existing Regulations

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 21 Chapter 4 Subchapter 1 Procedure and Conduct of Business of the California	Apr. 81	June 81	Public Meetings Notice Register Mailing Lists from: California Transportation Commission County Supervisors	The Commission intends to link the regulation review process to its regularly scheduled meetings throughout the state.	Air Resources Board Energy Commission California Highway Patrol Department of Motor Vehicles
Transportation Commission Subchapter 2 Guidelines for the Preparation of the State and Regional Transportation Improvement Programs (TIPS)		March 81	Association of California League of California Cities Newspapers Publications		Governor's Office of Planning and Research
Subchapter 3 Resolution G-14	Apr. 81	June 81			
Subchapter 4 Allocation of Funds for Inter- modal Interface Projects	Apr. 81	June 81			·
		,			

AGENCY: The California State University and Colleges

AGENCY COORDINATOR: Bruce Richardson

ADDRESS: 400 Golden Shore, Long Beach, CA 90802

TELEPHONE NUMBER: (213) 590-5618

OAL Master Plan For Review of Existing Regulations

Page <u>177</u>

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 5				·	
Part V				No. of the Control of	
Chapter 1					
Subchapter 1 Definitions	June 81	July 81	Mailing List	Final regulatory changes will be considered at	California Community Colleges
Subchapter 2 Educational Pro- gram	June 81	July 81		meetings of the Board of Trustees to be held March 24-25, April 22 and June 2-3. Board meetings for F.Y. 81-82 are not scheduled until May. However, they are usually held bimonthly.	California Postsecond- ary Education Commission.
Subchapter 3 Admission Require- ments	June 81	July 81			
Subchapter 4 Student Affairs	June 81	July 81			
Subchapter 5 Administration (§40000-42007)	June 81	July 81			·
Subchapter 5 Administration	July 81	Aug. 81			
Subchapter 6 Auxiliary Organ- izations	July 81	Aug. 81			·
Subchapter 7 Employees (§42008-43501)	July 81	Aug. 81	•		
			•		

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 7 Employees	Sept. 81	Oct. 81			
Subchapter 8 Environmental Quality	Sept. 81	Oct. 81			
Quality (843520-43681)				·	
		·			
		·			

AGENCY: Department of Education

AGENCY COORDINATOR: Donald R. McKinley, Chief Deputy Superintendent

ADDRESS: 721 Capitol Mall, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-5272

Page 179

No Plan Submitted

4950m, which also as subage of the transport of the control of the		T	NO FIRM Submitted		
ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 5				·	
Part I					
Division 1 General Provisions Division 2 Pupils					
Division 3 Handicapped Children Division 4					•
Gifted and Talented Pupil Program Division 5 Consolidated Cate-					
gorical Aid Program Division 5.2 Improvement of					
Elementary and Secondary Education Division 6 Certificated					
Personnel Division 7 Credentials and Institutions Pre-					
paring Candidates for Credentials Division 9					
Instructional Material		·			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Division 10					·
Courses of Study					
Division 10.2			İ		
Adult Education				-	
Division 11					
Special Programs Division 12				•	
Compensatory					
Education		ľ			
Division 13					
School Facilities		*			
and Equipment					
Division 14	!			•	
School Finance				•	
Division 15					
Child Nutrition '					
Programs					
Division 16					
Governing Boards					
of School Districts					
Division 17					
County Superinten- dents					
Division 18)		
California Schools					
for the Deaf,					
Blind or Neurolog-					
ically Handicapped					
Division 19					
Children's Develop					
ment Centers			1		
Division 20			1		
State Board of					
Education Proce-	·				
dures			·		
		1	<i>'</i>	্	

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Division 21 Private Postsecond- ary Education Institutions Division 22 Veterans Education			`.		•
Division 23 Miscellaneous Provisions					

Page <u>182</u>

AGENCY: State Controller's Office

AGENCY COORDINATOR: Bob Sparks

ADDRESS: 1227 O Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-9103

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 2 Chapter 2 Subchapter 1 Accounting Procedures for Counties Subchapter 2 Budgeting Procedures for Counties	June 81 Aug. 81	Aug 81 Oct. 81	Mailing List Sacramento Daily Recorder Controller's Advisory Committee on: County Accounting Procedure, Local Government Financial Reporting, County Tax Collecting Procedures	During review process, public meetings will be conducted with the advisory committees, with advisory committee making recommendations to the Controller's staff.	The State Controller's Division of Local Government Fiscal Affairs will contact and inform agencies which utilize or make reference to the regulations either by way of an announcement letter or personal contact.
Subchapter 4 Budgeting Procedures for Special Districts	Aug. 81	Oct. 81			
Subchapter 5 Minimum Audit Requirements and Reporting Guide- lines for Special Districts	Aug. 81	Oct. 81			
Subchapter 6 Tax Collection Procedures for Counties	June 81	Sept. 81			

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 7 City Financial Reports	Aug. 81	Oct. 81			
Subchapter 8 Unclaimed Property Law	July 81	Sept. 81			
Subchapter 3 Accounting Procedures for Special Districts					
Articles 1-4	May 81	July 81			
Articles 5-7	June 81	Aug. 81			•
Articles 8-10	July 81	Sept. 81	ļ		
Title 18					
Chapter 2.5					·
Subchapter l Inheritance Tax	June 81	July 81			
Subchapter 2	June 81	July 81			
Gift Tax					
					,
				,	
			·		

AGENCY: Department of Justice

AGENCY COORDINATOR: Roger Kramer

ADDRESS: 1315 5th St., Rm. 416, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 323-2191

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 11					
Chapter 1 Subchapter 1 Proceedings - Quo Warranto	Apr. 81	June 81	Newspapers Mailing List Questionnaires Trade Journals	Subchapter 7 - Criminal Offender Record Information Security - question- naire will be developed and distributed to all	Subchapter 2 - Approved Surety Bond Forms will be coor- dinated with each state agency that
Subchapter 3 Microfilming	Apr. 81	June 81	California Pharmacist Pharmacy West	police departments, sheriff's departments,	submits bond forms for approval.
Subchapter 4 Trustees for Charitable Purposes	Apr. 81	June 81	Public Meeting	district attorneys' offices and probation departments. An advisory committee consisting of	Coordination is also planned with the Department of Food
Subchapter 4.5 Management of Institutional Funds	Apr. 81	June 81		members of C.L.E.A.R.S., a professional law en- forcement records managers association, will be	and Agriculture and the Department of Health Services.
Subchapter 7 Criminal Offender Record Information Security	Apr. 81	June 81		formed. Subchapter 8 - Controlled Substance - representative groups of controlled sub-	·
Subchapter 9 Report of Child Abuse Article 2	Apr. 81	June 81	·	stance wholesalers and retailers will be contacted to solicit comments. Additional comments will be solicited through	•
Subchapter 13 Conflict of Interest Code	Apr. 81	June 81		notice in trade journals of general circulation.	
•				· ·	

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 15 Nonprofit Corpor- ation Law	Apr. 81	June 81		Subchapter 11 - Body Armor - principal manufacturers of Body Armor will be solicit-	
Subchapter 2 Approved Surety Bond Forms	May 81	June 81	×	ed to provide comments. Additional notice will be provided through Police Chief a journal of general circu-	
Subchapter 5 Health Care Ser- vice Plans	May 81	June 81		lation among manufacturers and vendors of law enforce- ment equipment.	
Subchapter 6 Election Law Civil Suits	May 81	June 81		Subchapter 12 - Tear Gas Weapons - notice of review will be sent to tear gas manufacturers, vendors,	
Subchapter 8 Controlled Substances	May 81	June 81		testing laboratories, sher- iff departments and police departments. A public meet-	
Subchapter 9 Report of Child Abuse Article 1	May 81	June 81		ing will be held in April at the Dept. of Justice Training Center, Sacramento Subchapter 14 - Forensic Alcohol and Drug Analysis -	
Subchapter 11 Police Body Armor	May 81	June 81		review notices will be sent to California Counties and	
Subchapter 12 Tear Gas Weapons	May 81	June 81		local Boards of Supervisors to solicit comments.	
Subchapter 14 Forensic Alcohol and Drug Analysis	May 81	June 81		Notice of review of existing regulations for all the subchapters will be done through the use of general circulation newspapers.	

AGENCY: Secretary of State

AGENCY COORDINATOR: Richard B. Maness

ADDRESS: 1230 J Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 445-0820

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 7 Chapter 1 Voter Registration Chapter 4 Punch Cards and Ballot Paper Chapter 9 Miscellaneous	Apr. 81 Apr. 81 Apr. 81	May 81 May 81 May 81	The following will be involved in the review process: Community Constituent Liaison Committees State Liaison Committee of the County Clerks Association Notary Public Liaison Committee Corporations Liaison Committee Uniform Commercial Liaison Committee Notices will be published in the Sacramento Union March 1981.	Public comments solicited in writing on or before 10:00 a.m. Friday, April 24, 1981 to: Office of the Secretary of State 1230 J Street, Room 232 Sacramento, CA 95814 ATTN: Richard B. Maness For Liaison Committee meeting places, dates and times contact Coordinator.)

AGENCY: State Treasurer

AGENCY COORDINATOR: Linda Brooks

ADDRESS: 915 Capitol Mall, Rm. 110, Sacto, CA 95814

TELEPHONE NUMBER: (916) 445-2563

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 2 Division 2			N	Dublic Mostines to detain	State Department of
Chapter 4			Newspapers: The Daily Recorder	Public Meetings to deter- mine regulation changes	Banking
Subchapter 2 State Deposits	Aug. 81	Nov. 81	The Intercity Express L.A. Daily Journal The Recorder	will be held during August 1981.	State Savings and Loan Commissioner
Subchapter 3 Value of Promis- sory Notes Pledg- ed to Secure Local Agency	Aug. 81	Nov. 81	Daily Commercial News The Bond Buyer Public Meetings Issue Papers		
Deposits					•
					,

AGENCY: State Treasurer - Districts Securities Division

AGENCY COORDINATOR: Violet Enander

ADDRESS: 120 Montgomery Street, Rm. 1025, San Francisco, CA 94104

TELEPHONE NUMBER: (415) 557-1932

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 10					·
Chapter 4			Mailing Lists	Final regulation changes	Department of Water Resources
Subchapter 1 General	Aug. 81	Jan. 82	Newsletter - "ACWA News" Advisory Commission	will be determined at a public hearing before the Districts Securities Ad-	Resources
Subchapter 2 Districts Secur- ities Law Provi- sions	Aug. 81	Jan. 82		visory Commission on November 20, 1981.	
Subchapter 3 Irrigation Districts	Aug. 81	Jan. 82			
Subchapter 4 California Water Districts	Aug. 81	Jan. 82			·
Subchapter 5 County Water Districts	Aug. 81	Jan. 82		•	
Subchapter 6 Water Storage Districts	Aug. 81	Jan. 82			•
Subchapter 7 District Securities Investigation	Aug. 81	Jan. 82			
Subchapter 8 County or City Reclamation of Land Provisions	Aug. 81	Jan. 82			. "

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Subchapter 9 Other Public Agencies	Aug. 81	Jan. 82			
Ageneres					
					• •
		·		ó	
				:	
					•

AGENCY: California Pollution Control Financing Authority

AGENCY COORDINATOR: Keith M. Seegmiller

ADDRESS: 915 Capitol Mall, Rm. 110, Sacto, CA 95814

TELEPHONE NUMBER: (916) 445-9597

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DA	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 4					
Chapter 11			Newspapers	Public Hearings to consider	
Article 1 General	Nov. 81	Jan. 82	Public Meetings Small Business Syndicate composed of Kidder-Peabody,	regulatory changes have been scheduled for August 25, 1981 at 2:00 p.m. in	Water Quality Control Board Solid Waste Management
Article 2 Definitions	Nov. 81	Jan. 82	Dean Witter, Bank of America, Blyth Eastman,	the State Treasurer's Office, 915 Capitol Mall,	Board
Article 3 Procedures Relating to Authority Officers and Mem- bers	Nov. 81	Jan. 82	Paine Webber and E.F. Hutton Bond Counsel-O'Melveny Myers, and Orrick, Har- rington, Sutcliffe	Room 110. If additional discussion is necessary, a hearing will be held on November 24, 1981.	
Article 4 General Provisions Relationg to Auth- ority Actions	Nov. 81	Jan. 82			
Article 5 Provisions Relating to Small Business Financings	Nov. 81	Jan. 82		·	
			·		
,					
					·
				·	

AGENCY: Office of Emergency Services

AGENCY COORDINATOR: Albert Lockhart

ADDRESS: 2800 Meadowview Road, Box 9577, Sacto, CA 95823

TELEPHONE NUMBER: (916) 427-4203

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 19 Chapter 2 Subchapter 1 Emergencies and Major Disasters	May 81	June 81	Written comments solicited from 400 cities and 58 counties involved in emergency services program. Newspapers Mailing Lists	General circulation news- papers in Redding, San Francisco, Fresno and Los Angeles will be used for notices. Mailings with regulation review infor- mation will be sent out in early April 1981.	Department of Social Services
				in curry inprint 25000	
		·			S Post
					UNIVERSITY OF THE PROPERTY OF

AGENCY: Office of Planning and Research

AGENCY COORDINATOR: Barry Steiner

ADDRESS: 1400 10th Street, Sacramento, CA 95814

TELEPHONE NUMBER: (916) 322-2218

ADMINISTRATIVE CODE UNDER REVIEW	PUBLIC COMMENT ENDING DATE	STATEMENT OF COMPLETION	METHODS OF PUBLIC INVOLVEMENT	REVIEW PROCESS	COORDINATION
Title 14 Division 6.3 Chapter 1 Regulations for Designation of Lead Agency for the Preparation of Environmental Documents	May 81	Jan. 82	Mailing Lists Newsletters Public Meeting	Office of Planning and Research and the Resources Agency will conduct a joint review of California En- vironmental Quality Act regulations.	Resources Agency
			·		
		* .			